REPORT DOCUM	ENTATION PAGE		OMB No. 0704-0188
Public reporting burden for this oblication of information is estimated to average 1 hour per response, including the time for reviewing instruction, maintaining the data needed, and completing and reviewing the collection of information. Send comments recarding this burden estimate or any			s, searching existing data sources, gathering and other aspect of this collection of information
including suggestions for reducing the burden	to Washington Headquarters Services, Directorate	for Information Operations and Reports, 1215 Je	fisrson Davis Highway, Suite 1024, Arlington,
A ACENICY LICE ONLY	ant and Budget, Paperwork Reduction Project (97	04-0188). Washington, DC 20503	
1. Adenci USE UNLI	2. REPORT DATE	Binal Technical	VERED. 1 .Tul 92-31 May 9/
		rinai iecinicar,	
4. TITLE AND SUBTITLE			5. FUNDING NUMBERS
Development of I	Process Technology	y for the	8786/02
Fabrication of I	8786/01		
		****	61101E
5. AUTHOR(S)			
Thomas W. Sigmon	n, Principal Inves	stigator	
-	-	AFOSR-T	895-0407
PERFORMING ORGANIZATION	NAME(S) AND ADDRESS(ES)		9.PERFORMING ORGANIZATION
Oregon Graduate	Institute of Scie	ence & Technology	REPORT NUMBER
20000 NW Walker	Rđ.		
Portland, OR	97291-1000		
	·		<b>h</b>
. SPONSORING/MONITORING AG	ENCY NAME(S) AND ADDRESS(ES	MTIC.	10 SPONSORING/MONITORING
AFOSR/NE			AGENCY REPORT NUMBER
110 Duncan Ave.	, Suite B115	I COELEUIE	
Bolling AFB, D.(	20332-0001	CUN 115119951	F49620-92-J-0396
1. SUPPLEMENTARY NOTES	<u></u>	B	
		l D	
			<b>,</b>
2a. DISTRIBUTION/AVAILABILIT	Y STATEMENT		12b. DISTRIBUTION CODE
APPROVED FOR PURLIC	RELEASE: DISTRIBUTIO	N UNLIMITED	
3. ABSTRACT (Maximum 200 v	vords)		
The development	of the measure	for the second second second second	
Si Gei - structure	of the processes necessary	101 Internation of Sub-10	Unm neteroepitaxial
planar, and definal	ble by available natterning	techniques (e.g. Y-ray e	-beam ion beam or
optical lithograph	y). The report describes	the investigation of two c	andidate natterning
processes, enhance	ed ion beam etching, and	standard e-beam lift off. 7	The formation of the
heteroepitaxial str	uctures, in the form of r	ano-wires, is carried out	using pulsed laser
induced epitaxy, a	in ultra-r pid (<100ns) er	pitaxial growth technique	driven by a pulsed
excimer laser. Development of a novel, high yield cross sectioning process to examine the			
wire structures wa	s also carried out during th	he course of this effort. The	his process used ion
beam deposition o	f a platinum etch mask, f	followed by an enhanced,	localized ion beam
etch to define the v	vire specimen. Results fro	om cross sectional transmi	ssion electron beam
analysis indicate s	uccessful formation of th	e heteroepitaxial wires wi	th no defects being
observed			-
observeu.			
	DTIC	QUALITY INSPECTED 3	······································
A. SUBJECT TERMS	DTIC	QUALITY INSPECTED 3	15. NUMBER OF PAGES
I. SUBJECT TERMS	DTIC	QUALITY INSPECTED 3	15. NUMBER OF PAGES 89
. SUBJECT TERMS	DTIC	QUALITY INSPECTED 3	15. NUMBER OF PAGES 89 16. PRICE CODE
I. SUBJECT TERMS	DTIC	QUALITY INSPECTED 3	15. NUMBER OF PAGES 89 16. PRICE CODE
. SECURITY CLASSIF-	DTIC (	QUALITY INSPECTED 3	15. NUMBER OF PAGES 89 16. PRICE CODE 20. LIMITATION OF ABSTRACT
SUBJECT TERMS	DTIC ( 18. SECURITY CLASSIF- ICATION	QUALITY INSPECTED 3 19. SECURITY CLASSIF- ICATION OF ABSTRACT	15. NUMBER OF PAGES 89 16. PRICE CODE 20. LIMITATION OF ABSTRACT

•

FINAL TECHNICAL REPORT FOR THE PERIOD JULY 1, 1992  $\rightarrow$  May 31, 1995

# DEVELOPMENT OF PROCESS TECHNOLOGY FOR THE FABRICATION OF MESOSCOPIC DEVICES

ARPA ORDER NO: 8786 PROGRAM CODE:

#### **OREGON GRADUATE INSTITUTE**

DATE OF CONTRACT: 7/1/92 DATE OF EXPIRATION: 6/30/95 CONTRACT AMOUNT: \$759,196 AFOSR CONTRACT NO.: F49620-92-J-0396

> THOMAS W. SIGMON PRINCIPAL INVESTIGATOR 415-723-0426 SIGMON@EE.STANFORD.EDU

DR. JANE A. ALEXANDER PROGRAM MANAGER-ARPA

DR. GERALD WITT-AFOSR AGENT TECHNICAL CONTACT

SPONSORED BY ADVANCED RESEARCH PROJECTS AGENCY ARPA ORDER NO. 6860 MONITORED BY AFOSR UNDER CONTRACT NO.: F49620-92-J-0396

THE VIEWS AND CONCLUSIONS CONTINED IN THIS DOCUMENT ARE THOSE OF THE AUTHORS AND SHOULD NOT BE INTERPRETED AS NECESSARILY REPRESENTING THE OFFICIAL POLICIES OF ENDORSEMENTS, EITHER EXPRESS OR IMPLIED OF THE THE ADVANCE RESEARCH PROJECTS AGENCY OR THE U.S. GOVERNMENT.

### **EXECUTIVE SUMMARY**

Nanometer-scale devices have attracted great attention as the ultimate evolution of silicon integrated circuit technology. However, fabrication of nanometer-scale silicon based devices has met great difficulty because it places severe constraints on process technology. This is especially true for SiGe/Si heterostructures because they are particularly sensitive to strain relaxation and/or process induced defects. Recently developed Pulsed Laser Induced Epitaxy (PLIE) offers a promising approach for the fabrication of nanometer-scale SiGe/Si devices. It possesses the advantage of ultra-short time, low thermal budget and full compatibility with current silicon technology. The selective nature of the process allows epitaxial growth of high quality, localized SiGe layers in silicon. In this work, a process to fabricate SiGe nanowires in silicon using PLIE is described. In particular, Ge nanowires with a cross-section of ~ 6 x 60 nm<sup>2</sup> are first formed using a lift-off process on the silicon substrate with e-beam lithography, followed by a thin low-temperature oxide deposition. Defect-free SiGe nanowires with a cross-section of ~ 25 x 95  $\text{nm}^2$  are then produced by impinging the laser beam on the sample. We thus demonstrate PLIE is a suitable fabrication technique for SiGe/Si nanostructures. Fabrication of Ge nanowires is also studied using Focused Ion Beam (FIB) micromachining techniques. Based on the SiGe nanowire process, we propose two advanced device structures, a quantum wire MOSFET and a lateral SiGe Heterojunction Bipolar Transistor (HBT). MEDICI simulation of the lateral SiGe HBT demonstrates high performance of the device. In order to characterize the SiGe nanowires using cross-sectional transmission electron microscopy, an advanced versatile focused ion beam assisted sample preparation technique using a multi-layer stack scheme for localized surface structures is developed and described in this work.



For

τ

1

1.00

# TABLE OF CONTENTS

Chapter 1 - Introduction1
1.1 Motivation for Work1
1.2 Organization
1.3 References4
Chapter 2 - PLIE System Set-up and Characterization7
2.1 Introduction7
2.2 Pulsed XeCl Excimer Laser System7
2.3 Fundamental Principles of the PLIE Process11
2.4 System Characterization15
2.5 References
Chapter 3 - Applications of SiGe Nanowires20
3.1 Introduction20
3.2 SiGe Quantum Wires20
3.3 SiGe Lateral HBT22
3.3.1 A Review of Lateral Bipolar Transistors
3.3.2 Design and Simulation25
3.4 References
Chapter 4 - Ge Nanowire Fabrication Using Focused Ion Beam Micromachining35
4.1 Introduction
4.2 Experimental Procedure
4.3 Results and Discussion
4.4 Conclusion and Future Work45
4.5 Reference
Chapter 5 - Ge Nanowire Fabrication Using E-beam Lithography
5.1 Introduction49
5.2 Experiment

5.3 Characterization and XTEM Sample Preparation	52
5.4 Results and Discussion	61
5.5 Conclusion	64
5.6 Reference	64
Chapter 6 - Fabrication of SiGe Nanowires Using Pulsed Laser Induced Epitaxy	68
6.1 Introduction	68
6.2 Formation of SiGe wires without LTO capping layer	70
6.3 Formation of SiGe wires using an LTO capping layer	78
6.4 Conclusion	84
6.5 Reference	85
Chapter 7 - Conclusions and Future Work	86
7.1 Summary and Conclusions	86
7.2 Suggestions for Future Work	88

.

# LIST OF TABLES

3.1	State-of-art bipolar transistors	24
-----	----------------------------------	----

# LIST OF FIGURES

2.1	Schematic Illustration of the pulsed XeCl excimer laser processing system. This system is controlled by a computer and includes in-situ, digitized diagnostic capabilities for real-time process monitoring
2.2	Simulation results of the reflectivity of a beam splitter for a 308 nm laser beam with different polarization condition at different angles. The beam splitter has a nominal reflectivity of 23(±5)%. Courtesy of Acton Research Corporation, Acton, MA 0172
2.3	Schematic process flow to illustrate the PLIE process using heteroepitaxial $Ge_xSi_{1-x}/Si(100)$ formation as an example. (a) A thin layer of amorphous Ge film is deposited onto a Si(100) substrate by e-beam evaporation. (b) Pulsed XeCl excimer laser melts through the amorphous layer and part of the substrate to form an epitaxial layer of $Ge_xSi_{1-x}$
2.4	Simulated melt/re-solidification process using the LASERMELT simulator for a Si(100) substrate during one laser irradiation pulse at an energy level of 0.82 J/cm <sup>2</sup> . The inset shows the digitized laser pulse used for the time domain calculation
2.5	TEM image of SiGe formed by 1 laser irradiation on a 7 nm Ge/Si substrate14
2.6	Results of spreading resistance profilometry (SRP) conducted on p <sup>+</sup> n junctions formed by gas immersion laser doping process. Ten laser pulses are used for each sample. The associated surface melt durations for each sample are indicated on the figure
2.7	<ul><li>(a) LASERMELT simulation results for experimental data.</li><li>(b) LASERMELT simulation showing the relationship of melt depth and laser energy fluence.</li></ul>
3.1	Schematic of the density of states versus energy for the semiconductor with no confinement, 1D confinement, and 2D confinement
3.2	Schematic of the quantum wire device (a) top view and (b) cross-section along A-B
3.3	Schematic illustration of the process flow (a)-(f), using Al/SiO2-masking (■ Al, □ SiO2) and the layout (g). Current flow direction is shown in the cross-section view of the final device (f)
3.4	Cross-sectional view of the SiGe lateral HBT (right half of the cylindrical

	structure). The dotted lines are current flow lines when $Vce = 3.0 V$ and $Vbe = 0.82 V$ . The dashed lines indicate the contact region
3.5	Doping profile along the surface of the SiGe lateral HBT with a base width of 0.1 $\mu$ m
3.6	Simulation results of the gummel plot for the SiGe lateral HBT with a base width of 0.05 $\mu$ m
3.7	Simulation results of Current gain v.s. collector current characteristics at three different base widths of the SiGe lateral HBT
3.8	Simulation results of cut-off frequency (Ft) and collector current (Ic)characteristics at three different base widths of the SiGe lateral HBT
4.1	FEI FIB 610 workstation
4.2	Cross-section of the ion column and sample stage in the vacuum chamber37
4.3	Illustration of the FIB milling process (a-b) and milling pattern (c) created from a start and end point file. The shaded area are defined to be milled in (c)
4.4	SEM image of the features milled (a)without using enhanced etch and (b) using enhanced etch
4.5	3-D AFM image of the feature milled using enhanced etch
4.6	(a) Schematics of the wedge shape. The slope of the wedge is about $3^{\circ}$ . (b) The shape of the feature expected from FIB milling with the presence of iodine if a milling depth of 100 Å is required and the center of the feature is not milled. The dotted box shows the ideal Ge bar if no tail effect of the FIB is involved
4.7	The calculated current density distribution for a FIB system at two different states of focus, one minimizing the rise distance of the beam across a knife edge and one optimizing the spatial resolution [4.20-4.21]
4.8	SEM image of Ge wire array fabricated by FIB micromachining from a 50 Å Ge/Si sample using iodine as enhanced etch agent. The arrows point to the droplets randomly scattered over the milling area. EDX shows a slight amount of iodine in the droplet
4.9	XTEM image of (a) the wedge at one side of the Ge wire fabricated using a FIB dose of $1.8 \times 10^{14}$ cm <sup>-2</sup> and (b) a sample received a FIB dose of $1.12 \times 10^{14}$ cm <sup>-2</sup> . The Ge film thickness before FIB milling is 50 Å
5.1	Illustration of the lift-off process to fabricate Ge nanowires using e-beam

•

hography
----------

5.2	Illustration of the cross-sectional TEM sample preparation process using FIB micromachining. (a) Mechanically saw a slice from the wafer containing the area of interest. (b) Mechanically polish this slice to $> 10$ mm. (c) Mount the slice on a modified TEM grid and mill two trenches, leaving a strip of materials which contains the feature of interest. The grid is then mounted to a support
5.3	Scanning ion image of (a) the Pt bar deposited by FIB assisted CVD on the area of interest and (b) the foil made by milling two trenches and leaving a strip of materials in between. The Pt is visible in the image because of its distinguished contrast with the Si substrate. Traces of Pt on the top of the foil are used as an indication of the foil thickness in the milling process
5.4	Scanning ion image of the "T" marks made at the sides of Ge wire array. The images are obtained by scanning using FIB at the lowest current possible, e.g. 6 pA, to avoid modification of the sample surface
5.5	Multilayer protection scheme to protect the surface of the substrate in the TEM, sample preparation process using FIB milling, particularly when the feature of interest is on the surface. (a) A thin layer of C(~ 100 Å thick) is thermally evaporated on the surface. (b) Au-Pd alloy (~ 1200 Å thick) is deposited using plasma sputtering
5.6	Cross-sectional TEM image of a sample prepared using (a) spin-on-glass or (b) a thin film of carbon (~ 100 Å thick) as the diffusion barrier. The surface of the Si substrate of both samples becomes amorphous, indicating neither scheme is ideal to protect an area of interest at the surface. The surface of the Si substrate underneath a particle is protected in image (b)
5.7	Lattice image of a sample prepared using FIB milling with multilayer protection scheme. The foil made by this technique generally is believe to be wedge shaped, i.e. thinner at the top of the foil, thicker at the bottom. However, areas of lattice imaging quality in the foil can extend more than 1 mm from the top of the foil, as in the case for this sample
5.8	Field Emission SEM image of three groups of Ge wires processed with e-beam dosages of 1.0, 0.95, 0.9 nC/cm from left to right respectively. The wires are not continuous when the e-beam dose used is blow 0.95 nC/cm
5.9	High magnification field emission SEM image of Ge wires processed with e-beam dose of 1.0 nC/cm viewed at a tilt angle of 740. The wires are seen to be fairly straight in the image
5.10	Atomic force microscopy image of a Ge wire showing the topology of the Ge surface

5.11	Cross-sectional TEM image of a Ge wire on a Si substrate formed after a lift-off process using e-beam lithography63
6.1	Cross-sectional TEM image of SiGe wells formed in a PLIE process using an Al/SiO <sub>2</sub> mask. No side diffusion is observed since the substrate area beneath the mask is not melted in the PLIE process. (Ref. 6.1)
6.2	Illustration of a possible melting and diffusion process in a PLIE process by directly applying the laser beam on the Ge nanowire on Si
6.3	Scanning electron micrograph of a Ge wire array on Si before laser irradiation. 
6.4	Scanning electron micrograph of the Ge wire array shown in fig.6.3 after laser irradiation
6.5	Cross-sectional TEM image following one laser irradiation of (a) Ge wires/Si with 0.5 mm spacing (b) Ge wires/Si with 1.0 mm spacing
6.6	Atomic force microscopy on a wire sample having 1-mm spacing showing decreasing ripple heights with increasing number of laser irradiation74
6.7	Nomarski micrograph for a wire sample after laser irradiation. Trenches can be seen surrounded by ripples75
6.8	Illustration of the ripple formation process76
6.9	Auger electron spectrum of SiGe wires formed by directly applying the laser beam on Ge nanowires/Si in a pulsed laser induced epitaxy
6.10	The melt pulse from an LTO covered Si substrate when a laser pulse (LLNL laser) is applied. The LTO thickness is ~ 15 nm. The FWHM of the melt pulse is 26 ns.
6.11	SEM image of the SiGe wires with a thin LTO capping layer after laser irradiation
6.12	(a) Cross-sectional TEM image of a SiGe wire formed by PLIE using LTO as a capping layer. (b) illustration of the SiGe wire interface and its relative location with respect to the original Ge wire
6.13	Cross-sectional TEM image of a SiGe wire showing the non-uniformity effect of the LTO layer on Ge wires
6.14	(a) SEM and (b) XTEM image of a contaminated Ge nanowires/Si sample after

laser irradiation.	A highly defected layer is formed at the	e surface of the substrate.

#### Chapter 1

#### INTRODUCTION

#### **1.1 Motivation for Work**

\* Scaling issues for ULSI

Continued desire for increasing speed and packing densities of VLSI circuits has driven minimum dimensions of devices down to the half-micron scale and integration levels up to 33 million devices on a  $100 \text{ mm}^2$  chip in manufacturing. The critical dimensions of devices are also predicted to decrease below 300 nm during the next five years, reaching 150 nm in late nineties.

However, further decrease in dimensions beyond 100 nm might prove to be problematic because of physical limitations [1.1], such as i) thermal and quantum perturbation; ii) statistical errors in threshold voltage due to small number of impurities in a unit volume; iii) electron tunneling through dielectric films below 4 nm in thickness and iv) builtin p-n junction depletion regions on the order of 20 nm.

#### \* Solution 1: Bandgap Engineering - Heterojunction Devices

During the last two decades, heterojunction devices[2.2-2.4] have received considerable attention for achieving higher performance and/or obtaining novel optical and electrical properties beyond the capabilities of physical limitations imposed by homojunction devices. Heterojunctions supplement homojunctions by the use of multiple semiconductor materials within a single device. It is understood that in homojunctions, holes and electrons accumulate only in those regions defined by their parent donor and acceptor impurities. In the contrary, heterojunctions can be used to drive both carriers into the same volume, to separate carriers from the parent dopant atoms, or to form local accumulation and/or depletion regions, independent of doping.

The heterojunction offers a large array of device configurations when compare to

homojunctions and becomes the basis of the field of "bandgap engineering". The introduction of III-V heterostructures more than twenty years ago has resulted in the production of superior devices and found commercial applications in the high-end areas such as the millimeter-wave and optoelectronics industry. Nevertheless, it is estimated that heterojunction devices account for no more than a few percent of the current production of semiconductor devices. This result stems from the fact that the dominant semiconductor material, silicon, does not easily form heterostructures with other semiconductor materials.

#### \* Solution 2: Nanoscale Switching Devices Concept

Quantum devices [1.5], including mesoscopic devices, represent another family of the most promising alternative devices to supersede silicon VLSI devices. Researchers have already observed interesting physical phenomena such as quantized conductance, wave-like behavior and enhanced luminescence. However, fabrication of such devices has been of great difficulty. Also in the last decade, most of the experimental research has been limited to III-V materials.

#### \* Si compatible heterojunction technology - Si-Ge

Advances in the material growth technologies, specifically, in molecular beam epitaxy (MBE) and Ultra High Vacuum Chemical Vapor Deposition (UHV/CVD) have made possible the growth of psuodomophic epitaxial layers. Therefore, strained  $Ge_xSi_{1-x}/Si$  heterostructures have become a subject of active attention in the heterostructure family because of its compatibility with silicon [1.6]. Various devices [1.7-1.8], such as vertical Heterojunction Bipolar Transistors (HBTs) with cut-off frequency up to 113 GHz [1.9], infrared detectors, and high electron mobility transistor structures have been demonstrated.

#### \* Selective Epitaxy of Si-Ge

The non-selective nature of most hetero-epitaxial growth techniques has proven to be an obstacle for exploitation of  $Ge_xSi_{1-x}/Si$  heterostructures for a large family of lateral

confinement devices such as lateral HBTs, quantum wires and dots. Several efforts [1.10-1.11] have been made with either little or no success or redesign of the device structures. To fabricate the above  $Ge_xSi_{1-x}/Si$  devices, a new materials growth technology is needed.

Recently developed Pulsed Laser Induced Epitaxy (PLIE) [1.12-1.13] offers a very promising alternative for fabrication of devices which require precise lateral confinement. The process works on a die-by-die basis and is fully compatible with current silicon lithography technology. It possesses an advantage of ultrashort process time scale (<1  $\mu$ s) and the sample is exposed to elevated temperatures for short times (< 0.2  $\mu$ s), resulting in a low thermal budget. The process has also demonstrated reliable process control [1.14]. Using this technique, excellent quality strained Ge<sub>x</sub>Si<sub>1-x</sub>/Si heterostructures have been demonstrated.

Fabrication of  $Ge_xSi_{1-x}/Si$  wells in a Si substrate using PLIE has also been reported [1.15]. In this experiment, defect-free SiGe wells of 3 and 6-µms in width were formed after PLIE using either standard oxide/Al masking or lift-off patterning techniques, respectively. The wells reported showed well defined, abrupt SiGe/Si interfaces and rounded corners for the structure.

In this work, we use the PLIE process to demonstrate fabrication of ultra-narrow SiGe wires in Si down to the 100-nm scale. We also discuss the possible applications of this result in lateral HBTs and quantum wires.

#### 1.2 Organization

This report is divided into seven chapters, with each chapter covering a major topic. In Chapter 1 we discuss the motivation for the work and the organization of the report.

Chapter 2 discusses the fundamental concepts and physics of the PLIE process along with describing the experimental set-up and the corresponding characterization techniques.

Chapter 3 discusses possible applications for the SiGe wires. We also introduce the design of a novel lateral heterojunction bipolar transistor based on the successful fabrication of the wires using PLIE. We demonstrate the high performance behavior of this device using MEDICI, a 2-dimensional device simulator.

Chapter 4 describes a possible approach for fabricating the ultra-narrow Ge wires using focused ion beam micromachining.

Chapter 5 describes the fabrication process used for sub-100 nm Ge wires using electron beam lithography and lift-off techniques. In this Chapter, we also describe the techniques used to characterize the wires, including atomic force microscopy, scanning electron microscopy and cross-sectional transmission electron microscopy.

Chapter 6 describes the laser mixing process for the Ge wires and Si substrate, *i.e.* the epitaxial growth of the SiGe wires in the silicon substrate. We also discuss topology changes observed in the epitaxial process.

Chapter 7 discusses the contributions of this work, and suggestions for possible future work based on the results obtained to date.

#### **1.3 References**

- [1.1] D. A. Antoniadis, J. E. Chung, "Physics and technology of ultra short channel MOSFET devices", in International Electron Devices Meeting 1991. Technical Digest. 21(1991).
- [1.2] High-speed Semiconductor Devices, John Wiley & Sons, Inc, edited by S. M. Sze, 1990.
- [1.3] P. Asbeck, "Heterojunction bipolar transistors: status and directions", in Proceedings of the 1989 Bipolar Circuits and Technology Meeting, 65(1989).
- [1.4] F. Capasso, "Bandgap and interface engineering for advanced electronic and photonic devices", *Thin Solid Films*, Vol. 216, 59(1992).
- [1.5] Physics of Quantum Electron Devices, Springer-Verlag, edited by F. Capasso,

1990.

- [1.6] J.C. Bean, "Silicon-Based Semiconductor Heterostructures: Column IV Bandgap Engineering", Proceedings of the IEEE, Vol. 80 No. 4, 571(1992).
- [1.7] M. Arienzo, J. Comfort II, E.F. Crabbe, D.L. Harame, S.S. Iyer, V.P. Kesan, B.S. Meyerson, G.L. Patton, J.M.C. Stork, "SiGe heterojunctions: devices and applications", *Microelectronic Engineering*, Vol. 19, No. 1-4, 519(1992).
- T. P. Pearsall, "Silicon-germanium alloys and heterostructures: optical and electronic properties", CRC Critical Reviews in Solid State and Materials Sciences, Vol. 15, No. 6, 551(1989).
- [1.9] E. Crabbe, B. Meyerson, D. Harame, J. Stork, A. Megdanis, "113-GHz f<sub>t</sub> graded-base SiGe HBT's", *IEEE Trans. on Electron Devices*, vol.40, no.11, 2001(1993).
- [1.10] Y.S. Tang, C.D.W. Wilkinson, C.M. Sotomayor Torres, D.W.Smith, T.E. Whall and E.H.C. Parker, "Optical properties of Si/Si/sub 1-x/Ge/sub x/ heterostructure based wires", *Solid State Commun.* Vol. 85, No. 3, 199(1993).
- [1.11] N. Usami, T. Mine, S. Fukatsu and Y. Shiraki, "Fabrication of SiGe/Si quantum wire structures on a V-groove patterned Si substrate by gas-source Si molecular beam epitaxy", *Solid-State Electron*. Vol. 37, No. 3-4, 539(1994).
- [1.12] J.R. Abelson, T.W. Sigmon, Ki Bum Kim and K.H. Weiner, "Epitaxial Ge<sub>x</sub>Si<sub>1-x</sub> / Si(100) structures produced by pulsed laser mixing of evaporated Ge on Si(100) substrates", *Appl. Phys. Lett.* Vol. 52, No. 3, 230(1988).
- [1.13] K. -J. Kramer, Talwar, S.; Sigmon, T.W.; Weiner, K.H., "Crystallinity, strain, and thermal stability of heteroepitaxial Si<sub>1-x</sub>Ge<sub>x</sub>/Si (100) layers created using pulsed laser induced epitaxy", Appl. Phys. Lett., Vol. 61, No. 7, 769(1992).
- [1.14] P.G. Carey, K.H. Weiner, T.W. Sigmon, "A shallow junction submicrometer PMOS process without high temperature anneals", *IEEE Electron Device Lett.*, Vol. 9, No. 10, 542(1988).

[1.15] Yih Chang, J. Kramer, S.Y. Chou, T.W. Sigmon, A.F. Marshall, "Fabrication of patterned Ge<sub>x</sub>Si<sub>1-x</sub>/Si layers by pulsed laser induced epitaxy", *Appl. Phys. Lett.* Vol. 58, no.19, 2150(1991).

#### Chapter 2

# PULSED LASER INDUSED EPITAXY SYSTEM SET-UP AND CHARACTERIZATION

#### 2.1 Introduction

The PLIE system employed in this study was set-up at the Oregon Graduate Institute (OGI). The results described in Chapter 6 are mostly obtained using this system. A somewhat similar system exists at Lawrence Livermore National Laboratory (LLNL), and has also been used for some of the work discussed in Chapter 6. In this chapter, we describe only the system set-up at OGI. A rigorous description of the LLNL system can be found in references 2.1 and 2.2 for the interested reader.

#### 2.2 Pulsed XeCl Excimer Laser System

The PLIE system is designed around a pulsed XeCl excimer laser. It is better known as a Gas Immersion Laser Doping (GILD) system [2.1-2.2]. The equipment required for the two processes are essentially the same. As shown in Fig. 2.1, the following three process modules are incorporated into the system.

i) Excimer laser, beam splitters, beam shaping optics, gas cell, and X-Y stages;

ii) In situ diagnosis and real time data acquisition apparatus;

iii) Homogenizing optics.

This laser-melting apparatus is designed to allow determination of all parameters required for characterization of the melt/diffusion/recrystallization process *in-situ* and in *real-time*. It utilizes a Questek 2860 excimer laser operating at the XeCl wavelength of 308 nm (3.9 eV photon energy) with a pulse duration full width half maximum (FWHM) of 27 nanoseconds. The excimer laser beam is directed through a set of reflective beam

splitters which allow coarse adjustment ( $\Delta E \ge 5\%$ ) of the laser energy. Fine tuning of the laser energy is through the adjustment of the high voltage supplied to the discharge capacitors of the laser. After the beam splitters, the beam is directed into an optical integrator which is used to homogenize the input laser beam. The refined beam is focused to a final spot on the wafer whose size can be varied from  $3x3 \text{ mm}^2$  to  $1x1 \text{ cm}^2$ . The wafer is contained in a portable vacuum cell filled with the dopant gas for *in-situ* doping or an inert gas for melt-recrystallization experiments. The cell is mounted on a set of x-y stages and the



Figure 2.1 Schematic Illustration of the pulsed XeCl excimer laser processing system. This system is controlled by a computer and includes in-situ, digitized diagnostic capabilities for real-time process monitoring.

process is usually carried out on a step and repeat mode and die-by-die basis under computer control.

The real time process diagnostics consist of a HeNe laser reflectivity probe, Si PIN diode, Molectron J50 laser energy fluence probe, Coho high resolution CCD camera, high resolution video monitor, and a Tektronix 2410 2-channel digitizing oscilloscope. Together, this equipment is used to: 1) measure the duration for which the wafer surface remains molten, 2) monitor the temporal profile of the excimer laser beam, 3) measure the laser energy fluence, 4) visually inspect the processed areas, and 5) visually align the wafer for die-by-die processing. The surface melt duration is measured using the HeNe ( $\lambda$ =633 nm) laser focused on the silicon wafer and an avalanche photodiode (APD) to detect the reflected beam. This laser and the APD constitute a simple surface reflectivity probe. Since the HeNe is non-intrusive, the measurement can be made *in-situ*. The temporal intensity profile of the excimer laser beam is acquired using a Si pin photodetector placed behind one of the turning mirrors. The output of this detector and that of the APD are collected by the oscilloscope, which are in turn collected by the computer system controlling the X-Y stages and the laser.

The laser beam energy measurement is made by placing a beam splitter in the laser beam path following the other beam splitters at a small angle ( $\leq 5^{\circ}$ ) relative to the incoming laser beam, as shown in Fig. 2.1. After passing through several beam splitters, the laser beam becomes partially polarized. The reflectivity of a beam splitter varies for different partially polarized light because of the difference between the reflectivity for the "S" and "P" components of the light. This setup minimizes the difference of the reflectivity of the "S" and "P" components of the laser beam. As shown in Fig. 2.2, the smaller the angle of the laser beam with respect to the beam splitter, the less difference between the reflectivity of the "S" and "P" components. The J50 thermal detector and JD-2000 Joul meter are used to integrate the total energy of the reflected beam and transfer the data to the computer. With the help of the beam profiling tools described below, the data can be easily converted to the laser energy fluence put down on the sample by a constant attenuation factor depending on how much fraction of the beam energy is lost when the beam passes through the homogenizer and the turning mirrors. The thermal detector and the Joul meter constitute a laser energy probe, which can be used *in situ*, as are the HeNe reflectivity probe.

The final module incorporated into the laser system allows spatial beam profiling and precise laser energy fluence data conversion. The beam profiling system consists of a Cohu CID camera, a frame grabber and a PC interface card, an IBM PC, and analysis software obtained from Big Sky Corporation. The ability to profile the beam and insure its homogeneity is perhaps the most critical requirement for accurate characterization of the process. Without detailed knowledge of the spatial uniformity, size and energy of the beam described above, it is difficult to accurately determine the energy fluence. Further-



Fig. 2.2. Simulation results of the reflectivity of a beam splitter for a 308 nm laser beam with different polarization condition at different angles. The beam splitter has a nominal reflectivity of  $23(\pm 5)\%$ . Courtesy of Acton Research Corporation, Acton, MA 01720.



Fig. 2.3. Schematic process flow to illustrate the PLIE process using heteroepitaxial  $Ge_xSi_{1-x}/Si(100)$  formation as an example. (a) A thin layer of amorphous Ge film is deposited onto a Si(100) substrate by e-beam evaporation. (b) Pulsed XeCl excimer laser melts through the amorphous layer and part of the substrate to form an epitaxial layer of  $Ge_xSi_{1-x}$ .

more, nonuniformities of the beam intensity can cause erroneous results and create thermal gradients across the molten region which lead to breakup of the recrystallization front, resulting in the growth of highly defected layers. The typical homogenized beam has an intensity variation of  $\pm 3\%$  across the central 3 mm of the spot. We have used both materials and electrical measurements, combined with modeling, to characterize the melt-doping process.

## 2.3 Fundamental Principles of the PLIE Process

PLIE is essentially a technique developed from the previously reported pulsed laser annealing and recrystallization process [2.3-2.5]. Initial studies indicated that by

using a Q-switched pulsed ruby or Nd-YAG laser to melt through a thin layer of deposited amorphous silicon, either doped or undoped, epitaxial layers structurally identical in quality to the under lying single crystal silicon substrate could be fabricated. In this work, the Q-switched ruby or Nd-YAG laser is replaced by an excimer laser operating at a wavelength of 308 nm.

Fig. 2.3 illustrates the fundamental principals of the PLIE process using heteroepitaxial Ge<sub>x</sub>Si<sub>1-x</sub>/Si(100) formation as an example [2.6]. First, an amorphous thin film is deposited onto a semiconductor substrate by, in this study, e-beam evaporation. If desired, dopant atoms can be incorporated in and/or onto the deposited film at the same time. PLIE is somewhat insensitive to interfacial contamination existing at the amorphous layer/substrate interface. Therefore, we believe the CVD techniques such as plasma enhanced CVD (PECVD) and low-pressure CVD (LPCVD), generally used in present day manufacturing lines can also be used for the deposition step.

Once the amorphous layer is deposited, the wafer is transferred to the laser system. It is mounted in a gas cell which can be purged and filled with either inert gases or a gaseous dopant species. Areas of the wafer are irradiated by the pulsed XeCl excimer laser. The laser energy is controlled such that the deposited thin film as well as part of the substrate is melted through. The melt depths and compositions of the melted layer are controlled by two independent adjustable laser parameters i.e. energy fluence or melt duration, and number of pulses.

Fig 2.4. illustrates the melting/re-solidification process for a Si(100) substrate following a single pulse laser irradiation at an energy fluence of  $0.82 \text{ J/cm}^2$ . The melting starts at the surface of the substrate and gradually reaches a maximum velocity. The resolidification begins at the maximum melt depth and proceeds toward the surface as the latent heat is dissipated into the underlying cooler substrate.

For the case of a thin film Ge layer on a Si substrate, both the deposited Ge and any other deposited species will redistribute in the molten region. The distribution of each



Fig. 2.4. Simulated melt/re-solidification process using the LASERMELT simulator for a Si(100) substrate during one laser irradiation pulse at an energy level of 0.82 J/cm<sup>2</sup>. The inset shows the digitized laser pulse used for the time domain calculation.

atomic species in this molten layer is of concern since their redistribution behavior determines the final electrical properties of the layer, such as the bandgap and Fermi level. It has been reported by Gladush *et. al.* [2.7] that liquid-phase diffusion and convection may contribute to the redistribution of the atoms in such a molten layer. The strong convection currents, caused by lateral temperature gradient induced surface tension gradients, may produce more efficient mixing than is possible by liquid phase diffusion. In their work, the laser used was a scanned beam and created temperature gradients laterally to the beam axis as well as along the scan direction.

In our work, lateral temperature gradients are believed to be of minor importance because of the excellent laser beam uniformity achieved in our system. Hence, the dominant redistribution mechanism in the molten region in our process is liquid phase diffu-

sion. The diffusion coefficients for this process are typically on the order of  $10^{-4}$  cm<sup>2</sup>/sec for semiconductors [2.2]. The fast liquid phase diffusion i.e. mixing in our case results from several possible factors; such as vertical temperature gradients, thermal stress acting through the piezoelectric effects and photovoltaic effects [2.8]. These phenomena are observed for several material systems [2.9, 2.10]. In Fig. 2.5, we show a cross-sectional TEM image of Ge<sub>x</sub>Si<sub>1-x</sub>/Si grown by PLIE for a 50 ns surface melt duration. The distinctive contrast between the alloy region and the substrate suggests excellent composition uniformity in the once-molten layer, attributed to the fast liquid diffusion of Ge in the vertical direction in the Ge-Si material system and the use of only one laser pulse. This is one of the many factors which make PLIE a technique suitable for ultra-small dimension Ge<sub>x</sub>Si<sub>1-x</sub>/Si wire fabrication, particularly when using the lift-off technique where side diffusion is an issue, as we discuss later in Chapter 6.

The composition uniformity of the alloy region may be modified by using multiple irradiation pulses. Since each pulse results in a melt-solidification process, all species in the molten region have a chance to redistribute. Typically, the composition profile reaches



Fig. 2.5. TEM image of SiGe formed by 1 laser irradiation on a 7 nm Ge/Si substrate.

a quasi-equilibrium situation for the Ge-Si system after a few pulses [2.11]. Afterwards, segregation effects dominate the composition redistribution. A detailed review of this subject is presented in reference 2.2.

The PLIE process is essentially a very rapid liquid phase epitaxial process. The recrystallized crystal structure preserves the atomic arrangement of the substrate. In the case of heteroepitaxy, strained layers are formed, as observed in the  $Ge_xSi_{1-x}/Si$  system. In order to obtain high quality epitaxial layer, uniform epitaxial growth is essential. It has been previously reported that polycrystalline and cellular formation is resulted from the laser-induced recrystallization process [2.12]. However, the lasers used in that work, both continuous and pulsed types, are known to have significant spatially nonuniform beam intensities. These intensity variations create inhomogeneous melt front and result in random nucleation for the recrystallization process. This problem is resolved by using a uniform laser beam.

Although PLIE is a reliable fabrication technique, there exist several factors that are not fully understood, such as high temperature gradients, thermal stress and pressure exerted by the pulsed laser, as pointed out in reference 2.8. New phenomenon in the PLIE process are continually being discovered, such as surface evolution in the fabrication of ultra-small SiGe wires, which we discuss in Chapter 6. Future work is needed in order to establish a complete theoretical understanding of the mechanisms of the PLIE process.

#### 2.4 System Characterization

In order to precisely control the PLIE process for fabrication of ultra-narrow SiGe wires, the system needs be well characterized. The critical parameter to be monitored in the PLIE process is the melt depth. It determines the epilayer thickness and in turn determines the concentration of the species in the epilayer. This parameter is directly related to the laser energy fluence and may be measured by monitoring the surface melt duration.

The GILD process is performed on well characterized Si substrates and together



Fig. 2.6. Results of spreading resistance profilometry (SRP) conducted on  $p^+n$  junctions formed by gas immersion laser doping process. Ten laser pulses are used for each sample. The associated surface melt durations for each sample are indicated on the figure.

with the LASERMELT simulator to monitor the system. The GILD process is similar to the PLIE process, except for the fact that it basically is used to incorporate dopants in the surface layer through the gas phase rather than creating a change in the actual material through epilayer growth. Because of the nanosecond thermal cycles and the fast diffusion of dopants in the molten region, dopant atoms only diffuse up to the liquid/solid interface forming a box-like profile after multiple laser pulse irradiation; solid state diffusion during the ultra-fast heat cycle being negligible. The epitaxial regrowth occurring during the solidification process of the doped layer results in the impurities being incorporated onto substitutional lattice sites and these dopants are therefore electrically active.

Fig. 2.6 shows impurity profiles obtained using spreading resistance profilometry (SRP) on low-doped n(100) Si substrates processed by GILD in a BF<sub>3</sub> ambient. The BF<sub>3</sub>



Fig. 2.7. (a) LASERMELT simulation results for experimental data. (b) LASERMELT simulation showing the relationship of melt depth and laser energy fluence.

17

gas pressure in the gas cell before laser irradiation is 50 torr. Each sample is irradiated with ten pulses at different energy fluence by the homogenized laser beam. The experimental results are fit with the LASERMELT simulator, and are shown in Fig. 2.7(a). In this figure, a finer grid is used for the simulation marked "grid 2". The abrupt change of the curve between 120 - 130 ns is caused by the grid setting used in the calculation. Note, that the experimental results for a highly doped n(111) substrate (dopant concentration =  $1.5x10^{19}$  atoms/cm<sup>3</sup>) are also shown. They do not fit the simulated results since the properties of the highly doped silicon substrates are changed by doping. Fig. 2.7(b) is the plot of simulated results showing the relationship of melt depth and laser energy fluence.

#### 2.5 References

- [2.1] P.G. Carey, Ph.D. Dissertation, Stanford University, Stanford, CA, 1988.
- [2.2] K.H. Weiner, PH.D. Dissertation, Stanford University, Stanford, CA, 1989.
- [2.3] I.W. Boyd and J.I.B. Wilson, "Laser processing of silicon', Nature, 303, 481(1983).
- [2.4] R.F. Wood, C.W. White, and R.T. Young, "Laser processing of semiconductors: An overview", in Semiconductors and semimetals, edited by R.F. Wood, C.W. White, and R.T. Young, 23, 1(1984)
- [2.5] Laser annealing of semiconductors, Academic Press, edited by J.M. Poate and J.W. Mayer, 1982.
- [2.6] J.R. Abelson, T.W. Sigmon, K.B. Kim, and K.H. Weiner, "Epitaxial GexSi1-x/Si
- (100) structures produced by pulsed laser mixing of evaporated Ge on Si (100) substrates", Appl. Phys. Lett., 52(3), 230(1988).
- [2.7] G.G. Gladush, L.S. Krasitskaya, E.B. Levchenko, A.L. Chernyakov, Sov. J. Quantum Electron., 12, 408(1982).
- [2.8] Y. Chang, Ph.D. Dissertation, Stanford University, Stanford, CA, 1991.
- [2.9] Y. Chang and T.W. Sigmon, "Anomalous point defect injection during pulsed

laser melting processes: direct evidence of Ga/sub i/ and As/sub i/ profiles in GaAs", in *Phase Transformation Kinetics in Thin Films Symposium*, Mater. Res. Soc, 263(1991).

- [2.10] Y. Chang, S. Talwar, E.Y. Shu, T.W. Sigmon, "Direct evidence of the Si interstitial injection and fast diffusion effect in Si during pulsed laser melt process", in *Phase Transformation Kinetics in Thin Films Symposium*, Mater. Res. Soc., 257(1991).
- [2.11] J. Kramer, Ph.D. Dissertation, Stanford University, Stanford, CA, 1994.
- [2.12] M. von Allmen, "Laser-beam interactions with materials", Springer-Verlag, 1987.

#### Chapter 3

# **APPLICATIONS OF SIGE NANOWIRES**

#### **3.1 Introduction**

Because of the complexity involved in the processing techniques, there are no demonstrations of the application of SiGe wire structures to date. Most research activities are focused on the fabrication techniques, rather than electrical properties, even though it is of great interest.

In this chapter, we will discuss two applications of SiGe wires. One is quantum wire based and which we will propose the fabrication of a novel device. The other is a lateral heterojunction bipolar transistor (HBT) whose performance is demonstrated on a twodimensional device simulation tool - MEDICI.

#### 3.2 SiGe Quantum Wires

One application of SiGe wires is two dimensionally confined quantum devices. Coherently strained  $\text{Ge}_x\text{Si}_{1-x}$  has a smaller bandgap than Si. The bandgap [3.1] can be approximated by the relationship

$$E_{a}(x,T) = E_{0}(T) - 0.96x - 0.43x^{2} - 0.17x^{3}$$
 (3.1)

where T is the temperature, and  $E_0(T)$  is the bandgap of Si. Most of the bandgap offset is located at the valence band edge. The offset at the valance band can be approximated [3.2] by

$$\Delta E_{\rm m} = 0.74 \mathrm{x} \tag{(3.2)}$$

(A 1)

(2.2)

for strained  $Ge_{1-x}Si_x/Si$ . Therefore, holes will be driven into the alloy region. When a SiGe wire embedded in Si has a cross-section dimension smaller than the carrier deBrolie wavelength, the energy levels in the well are quantized, resulting in two-degreeless freedom of motion and forms the so-called "quantum wire".

As shown in Fig. 3.1, The density of states in such a 2-D confined quantum structure shows a significant discrete nature when compared to that of 1-D confined structure. Hence, tunneling behavior of the carriers, as seen in a resonant tunneling transistor, can be expected. In addition, an enhanced mobility along the quantum wire is possible [3.3], because of suppression of ionized-impurity scattering.

Fig. 3.2. shows a quantum wire MOSFET based on SiGe wires in Si. A thin undoped Si epi-layer is first grown on a p-type Si substrate followed by SiGe wire formation in the epi-layer. The electrical contacts can be formed on both ends of the wire utilizing the conventional metalizing technique. A gate structure (as an option) can be made over the wire with a thin deposited  $SiO_2$  in between. This gate structure could be used to change the transport behavior of the carriers through the wire, thus providing a three terminal device.



Fig. 3.1. Schematic of the density of states versus energy for the semiconductor with no confinement, 1D confinement, and 2D confinement.



Fig. 3.2. Schematic of the quantum wire device (a) top view and (b) cross-section along A-B.

#### 3.3 SiGe Lateral HBT

As discussed in 3.2, strained  $\text{Ge}_{1-x}\text{Si}_x$  has a bandgap narrower than Si. A bandgap difference in the order of 200 mev (8kT) can be obtained when the Ge mole fraction is about 20%. In a bipolar transistor, the current gain [3.4], as limited by emitter injection efficiency, can be approximated by

$$h_{fe} = \frac{n_E v_e N_{CB} N_{VB}}{p_B v_h N_{CE} N_{VE}} exp\left(\frac{\Delta E_g}{kT}\right)$$
(3.3)

where  $n_E$  is the emitter free electron concentration,  $p_B$  is the base free-hole concentration,  $N_{CB}$  and  $N_{CE}$  are the densities of states in the conduction band for the base and emitter,  $N_{VB}$  and  $N_{VE}$  are the densities of states in the valence band for the base and emitter,  $v_e$  and  $v_h$  are the effective velocities of electrons and holes, and  $\Delta E_g$  is the bandgap difference between the emitter and base, i.e.  $E_{gE} - E_{gB}$ . Therefore, incorporation of Ge into the base of the transistor will dramatically enhance the emitter injection efficiency in a NPN

BJT. This allows obtaining an increased current-gain cut-off frequency  $f_t$  which, in general, reflects the speed performance of the transistor, and/or increased maximum frequency of oscillation  $f_{max}$  [3.4] which is approximated by

$$f_{max} = \left(\frac{f_t}{8\pi R_B C_{BC}}\right)^{1/2}$$
(3.4)

where  $R_B$  is the base resistance,  $C_{BC}$  is the base-collector capacitance.

Since their first demonstration in 1987 [3.5, 3.6], SiGe HBTs have attracted great attention. Commercial application of these devices are a reality. However, all of the work to date has concentrated on vertical structures rather than lateral, due to a lack in the selective epitaxial growth technology for SiGe.

In this section, we present a brief review of current lateral bipolar junction transistor (BJT) technology, We then propose a novel NPN SiGe lateral HBT and its processing procedures based on PLIE. Finally, we discuss the performance of this device and a conclusion is given.

## 3.3.1 A Review of Lateral Bipolar Transistors

Lateral bipolar structures are commonplace in silicon technologies, but not well developed for high speed, high frequency applications. Table 3.1 lists the performance of the state-of-the-art lateral BJTs fabricated using different technologies. The renewed interest in high performance lateral BJTs are driven by the prospect of incorporating BJT and CMOS on the same chip using a simplified, CMOS like, BiCMOS technology [3.7] to exploit the high speed of CMOS and high power of BJTs at the same time. One solution would be using the parasitic BJTs [3.8,3.9] which are inherently available in a CMOS process. One could make use of either the vertical BJT which has the substrate as its collector or the lateral BJT between the source and drain junctions of the MOSFETs [3.10,3.11]. However, the vertical BJT can only be used in a common collector configuration since the collector is always tied to the substrate potential. The parasitic lateral BJT has a low cur-

rent gain, low current driving capability, and large base transit time.

Recently, S. Verdonckt-Vandebroek et al [3.12] demonstrated a high gain lateral BJT formed by tieing together the gate and well of a MOSFET. Such transistors have lower  $f_t$ -I<sub>c</sub> performance than conventional vertical BJTs and they are more like MOSFETs than BJTs, especially at high current. A. Tamba et al [3.13] demonstrated a lateral BJT using a modified MOS process. The cut-off frequency  $f_t$  reached 6.3 GHz and no buried layer is needed. In this device, the base resistance increases with decreasing base width and scaling of the device tends to be a problem. With the advent of SOI technology, several attempts [3.14-3.16] have been made to build lateral BJTs on the substrate. The highest  $f_t$  obtained using this technology is 20 GHz with a current gain of 20.

	High Performance Vertical BJT [3.17]	High Performance Lateral BJT [3.13]	Lateral BJT on SIMOX
Base width	1300 nm	~ 500 nm	~ 100 nm
β	72	20	10 ~ 30
F	25 GHz	6.3 GHz	20 GHz
 	306 Ω	450 Ω	
R_	20 Ω	50 Ω	
	10 Ω		8 Ω
 	11 fF		21 fF
<u> </u>	10 fF		32 fF
$C_{jb}$	1011		

Table 3.1: State-of-art bipolar transistors

There are few demonstrations of lateral HBTs. To date, using wet etching and regrowth techniques [3.18], lateral InP/InGaAsP HBTs with current gain of 6 and base width of 2  $\mu$ m were demonstrated. Using impurity induced disordering [3.19], GaAs/GaAlAs lateral HBTs with a current gain up to 400 were also demonstrated. The latter approach is better suited to the fabrication of monolithic laser/bipolar circuits. Neither author discussed the high frequency performance of the device. In addition, both

approaches are based on III-V materials. However, these approaches demonstrate the possibility of fabricating laterally structured HBTs.

In the following, we propose a novel lateral HBT design based on SiGe/Si heterojunctions. Simulation using MEDICI shows  $f_t$ 's could reach as high as 36 GHz with current gains of ~ 250 and a base width of 50 nm.

#### 3.3.2 Design and Simulation

Fig. 3.3. (f) and (g) shows the cross-section and layout of the device. The emitter is surrounded by the  $Si_{1-x}Ge_x$  base in order to make full use of the emitter area. Base contact is made through the buried layer which is partially overlapped by the SiGe base. A suggested process flow based on GILD and PLIE is shown in Fig. 3.3 (a)-(f). The GILD process is described in detail in ref. 3.20. Al/SiO<sub>2</sub> masks are used in the process for patterning and shielding the rest of the substrate from the laser beam. A boron-doped is first formed using the GILD technique in a selected area. Then a buried layer is formed in the p-well using ion implantation along appropriate channel direction followed by thermal annealing or using multiple laser irradiation [3.20, 3.21]. The edge area of the p-well is then further doped with boron for the base contact later on. The center area of the p-well is counterdoped with n-type donors to create an n-well in the p-well, as shown in Fig. 3.3(b). The SiGe base region is selected in the n-well and the SiGe alloy is formed using either lift-off techniques or regular lithography, followed by PLIE, as described in this report. If a liftoff technique is used, base dopant needs to be incorporated into Ge at the time Ge is deposited, or to be put down in a solid form together with the Ge. If regular lithography is used, the base dopant can be incorporated at the same time of PLIE using GILD. After stripping-off Al/SiO<sub>2</sub>, if there is any, a thin SiO<sub>2</sub> layer is redeposited and openings are made for contacts. Emitter contact is made through n<sup>+</sup> poly-Si, preferably. Base-collector capacitance is controlled by the position of collector contact relative to the location of the base region, or a lightly-doped poly-Si collector. The effect from the parasitic vertical BJT


Fig. 3.3. Schematic illustration of the process flow (a)-(f), using Al/SiO<sub>2</sub>-masking ( $\square$ Al,  $\square$ SiO<sub>2</sub>) and the layout (g). Current flow direction is shown in the cross-section view of the final device (f).

between the emitter and the substrate is suppressed by the highly doped buried layer. Except for the Ge incorporation step, the process is fully compatible with an advanced CMOS process.

Simulation of the device was performed using the MEDICI device simulator. MEDICI is a simulation program that can be used to model the two-dimensional distributions of potential and carrier concentrations in a device such as MOSFET or BJT to predict its electrical characteristics for any bias condition. It solves Poisson's equation and both the electron and hole current continuity equations to analyze devices and effects in which the current flow involves both carriers. AC small signal analysis at a given frequency can also be performed to calculate frequency dependant capacitances, conductances, and admittances. For the strained SiGe/Si, a piecewise linear approximation to the lower curve of the strain split data shown in Fig. 2 of [3.22] is implemented for the energy band alignment. The actual bandgap value of the strained Si<sub>1-x</sub>Ge<sub>x</sub> used in the MEDICI is converted from the data obtained in ref. [3.22], as shown below:

$$E_{g}(x) = E_{g}(90) - 4.0 (E_{g}(90) - 0.950) x, x \le 0.25$$
 (3.5)

Where Eg (90) is the bandgap value at a temperature of 90 K.

In the simulation, a cylindrical structure of the device is used instead of a square structure shown in Fig. 3.3(g). The right half of the cross-section of the cylindrical structure is shown in Fig. 3.4. The radius of the emitter region is chosen to be 0.1 micron. Box-like doping profiles were used for all the doped regions in the device. The depth of the emitter and collector was chosen to be 50 nm, which can be easily made using the GILD process [3.23]. The depth of the SiGe region was 100 nm [3.24] with constant concentration of Ge (20%) in the first half and linearly graded to zero in the second half. The doping profile along the surface of the device is shown in Fig. 3.5. Half of the emitter and collector was used for contact, as shown by the dashed line in Fig. 3.4. Current flow also shown in Fig. 3.5 is concentrated in the first half of the base region and unlike MOSFETs, almost uniformly distributed in this region.







Fig. 3.5. Doping profile along the surface of the SiGe lateral HBT with a base width of 0.1  $\mu m.$ 



Fig. 3.6. Simulation results of the gummel plot for the SiGe lateral HBT with a base width of 0.05  $\mu$ m.

Performance of the device is very similar to a vertical SiGe HBT. The Gummel plot [Fig. 3.6] of the device shows the typical characteristic of a BJT. The zero bias emitter junction capacitance is found to be 0.2 fF, while the collector junction capacitances are 0.42 fF, 0.56 fF and 0.84 fF corresponding to base width of 0.05  $\mu$ m, 0.1  $\mu$ m and 0.2  $\mu$ m, respectively. Base width scaling significantly enhances the performance of the device. As shown in Fig. 3.7 and 3.8, a 200 nm wide base gives current gain of 10 and maximum f<sub>t</sub> of 4.5 GHz. When the base width decreases to 50 nm, the maximum f<sub>t</sub> reaches 36 GHz with a current gain of 260.



SiGe LHBT - Beta vs. Ic

Fig. 3.7. Simulation results of Current gain v.s. collector current characteristics at three different base widths of the SiGe lateral HBT.

In summary, an advanced NPN lateral SiGe HBT is designed. Processing procedures fully compatible with advanced CMOS process with an additional mask for SiGe base formation is described for the device based on the GILD and PLIE fabrication technique. Simulation using a two-dimensional device simulator - MEDICI shows a maximum



Fig. 3.8. Simulation results of cut-off frequency (Ft) and collector current (Ic) characteristics at three different base widths of the SiGe lateral HBT.

cut-off frequency of 36 GHz with a current gain of 260 can be obtained. Performance characteristics of the device is comparable to vertical HBT structures. Future work includes optimization and demonstration of the device using GILD and PLIE processing.

#### **3.4 References**

- [3.1] D. V. Lang, R. People, J. C. Bean and A. M. Sergent, "Measurement of the band gap of Ge<sub>x</sub>Si<sub>1-x</sub> strained-layer heterostructures", *Appl. Phys. Lett.*, vol. 47, 1333(1985).
- [3.2] R. People, "Physics and applications of GrxSi1-x/Si strained-layer heterostructures", *IEEE J. of Quantum Electronics*, Vol. QE-22, No. 9, 1696(1986).
- [3.3] H. Sakaki, "Scattering suppression and high-mobility effect of size-quantized electrons in ultrafine semiconductor wire structures", JPN. J. Appl. Phys, Vol. 19, L-

735(1980).

- [3.4] P. M. Ashbeck, "Bipolar transistors", *High-Speed Semiconductor Devices*, Edited by S. M. Sze, John Wiley & Sons, Inc., 1990.
- [3.5] H. Temkin, "GexSi1-x optoelectronic devices", in Proc. 2nd Si-MBE Symp., Honolulu, HI. 28(1987).
- [3.6] S. S. Iyer, G. L. Patton, S. L. Delage, S. Tiwari, and J. M. C. Stork, "Silicon germanium base heterojunction bipolar transistors by molecular beam epitaxy", in *Proc. 2nd Si-MBE Symp.*, Honolulu, HI. 114(1987).
- [3.7] N. Akiyama, A. Tomba, Y. Wakui, and Y. Kobayashi, "CMOS-compatible lateral bipolar transistor for BiCMOS technology: part I - modeling", *IEEE Trans. Electron Devices*, vol. 39, 948(1992).
- [3.8] H. C. Lin, J. C. Ho, R. R. Iyer, and K. Kwong, "Complementary MOS-bipolar transistor structure", *IEEE Trans. Electron Devices*, vol. ED-16, 945(1969).
- [3.9] O. Minato, T. Masuhara, T. Sasaki, Y. Sakai, M. Kubo, K. Uchibori, and T. Yasui,
  "A high speed low power Hi CMOS 4K static RAM", *IEEE Trans. Electron Devices*, vol. ED-26, 882(1979).
- [3.10] S. S. Ahmed, W. W. Asakawa, M. T. Bohr, S. S. Chambers, T. Deeter, M. Den ham, J. K. Greason, W. W. Hoplt, R. R. Taylor, I. Young, "A triple diffused approach for high performance 0.8 µm BiCMOS technology", *Solid State Tech.*, Vol. 35, No. 10, 33(1992).
- [3.11] T.-W Pan and A.A. Abidi, "A 50 dB variable gain amplifier using parasitic bipolar transistors in CMOS", IEEE J. Solid-State Circuits, Vol. 24, 951(1989).
- [3.12] S. Verdonckt-Vandebroek, S. S. Wong, J. C. S. Woo, P. K. Ko, "High-gain lateral bipolar action in a MOSFET structure", *IEEE Trans. Electron Devices*, vol. 38, 2487(1991).
- [3.13] A. Tamba, T. Someya, T. Sakagami, N. Akiyama, and Y. Kobayashi, "CMOScompatible lateral bipolar transistor for BiCMOS technology. II. Experimental

results." IEEE Trans. Electron Devices, vol. 39, 1865(1992).

- [3.14] N. Higaki, T. Fukano, A. Fukuroda, T. Sugii, Y. Arimoto, T. Ito, "A 4 GHz thinbase lateral bipolar transistor fabricated on bonded SOI", *IEICE Trans. on Electronics*, vol.E75-C, no.12, 1453(1992).
- [3.15] G. G. Shahidi, D. D. Tang, B. Davari, Y. Taur, P. McFarland, K. Jenkins, D. Danner, M. Rodriguez, A. Megdanis, E. Petrillo, Polcar, "A novel high-performance lateral bipolar on SOI", in 1991 IEDM Tech. Dig., 663(1991).
- [3.16] S. A. Parke, C. Hu, P. K. Ko, "A high-performance lateral bipolar transistor fabricated on SIMOX", *IEEE Trans. Electron Devices*, vol. 14, 33(1993).
- [3.17] T.F. Meister, H.W. Meul, R. Stengl, D. Hartwig, R. Weyl, "A 26ps selective epitaxial bipolar technology", in 1991 Symposium on VLSI Technology, Digest of Technical Papers, Held: Oiso, Japan, 1991, P67.
- [3.18] H.-J. Yoo, J.R. Hayes, C. Caneau, R. Bhat, M. Koza, "Fabrication and characterization of lateral InP/InGaAsP heterojunctions and bipolar transistors", *Appl. Phys. Lett.*, Vol. 54, 2318(1989).
- [3.19] R. L. Thornton, W. J. Mosby, H. F. Chung, "Unified planar process for fabricating heterojunction bipolar transistors and buried-heterostructure lasers utilizing impurity-induced disordering", Appl. Phys. Lett., Vol. 53, 2669(1988).
- [3.20] P.G. Carey, Ph.D. Dissertation, Stanford University, Stanford, CA, 1988.
- [3.21] G. Verma, A. Slaoui, T. W. Sigmon, to be published.
- [3.22] S.S. Iyer, G.L. Patton, J.M.C. Stork, B.S. Myerson and D.L. Harame, "Heterojunction bipolar transistors using Si-Ge Alloys", *IEEE Trans. Electron Devices*, vol. 36, 2043(1989).
- [3.23] K. H. Weiner, P. G. Carey, A.M. McCarthy, T.W. Sigmon, "An excimer-laserbased nanosecond thermal diffusion technique for ultra-shallow pn junction fabrication", *Microelectronic Engineering*, Vol. 20, No. 1-2, 107(1993).

[3.24] K. -J. Kramer, S. Talwar, T.W. Sigmon, K.H. Weiner, "Crystallinity, strain, and thermal stability of heteroepitaxial Si<sub>1-x</sub>Ge<sub>x</sub>/Si (100) layers created using pulsed laser induced epitaxy", Appl. Phys. Lett., Vol. 61, No. 7, 769(1992).

## Chapter 4

# GE NANOWIRE FABRICATION USING FOCUSED ION BEAM MICROMACHINING

#### 4.1 Introduction

In order to epitaxially grow SiGe nanowires, a technology to fabricate ultra narrow Ge wires on a Si substrate needs to be developed. There are a number of existing lithography technologies, such as e-beam lithography, which will be addressed in the next chapter. Focused ion beam (FIB) micromachining, however, seems to be a simple approach to form Ge nanowires directly by selectively removing a thin film Ge layer deposited on Si.

FIB technology [4.1-4.9] is made possible by the development of field emission liquid metal ion sources and their incorporation into electrostatic ion optical focusing columns. These quasi-point sources provide scanning focused ion beams with diameters less than 50 nm. Since the 1980's, FIB technology has been increasingly used for a variety of purposes such as implantation, milling, surface chemistry, lithography, microscopy, and materials analysis [4.10-16]. Focused ion beam micromachining offers a very flexible and localized method of etching materials without the need or use of a separate patterned mask. Material removal is carried out by both milling with the focused ion beam alone or by gas enhanced etching, where the surface being bombarded by the ions is also exposed to reactive gas molecules [4.17-4.19].

In this chapter, we describe our investigation into the FIB milling of amorphous Ge thin films on Si. Two processes with and without using an enhanced etch compound (Iodine) are compared. The final structures are evaluated by atomic force microscopy (AFM), scanning electron microscopy (SEM), and cross-sectional transmission electron microscopy (XTEM). Possible improvements of the processes for future work are dis-



Fig. 4.1. FEI FIB 610 workstation

cussed.

# 4.2 Experimental Procedure

Our experiment employs an FEI FIB 610 workstation (Fig. 4.1). It has gas injection capabilities for iodine and a platinum compound. The configuration of the ion beam column and the sample stage in the vacuum chamber is shown in Fig. 4.2. For our milling experiments, the beam size of the gallium FIB is maintained around 280 Å, the beam current is kept at 6 pA with a beam energy of 25 keV. During milling, a dwell time of 1  $\mu$ s of the ion beam for each pixel is used and the beam overlap between adjacent pixels is 50%. Fig. 4.3(a)-(b) illustrates the process flow for the experiment. The sample used for the experiment is a n-type Si(100) wafer. Before FIB milling, a thin film of amorphous germanium, typically 50 ~ 100 Å thick, is e-beam deposited at room temperature under a vac-



Fig. 4.2. Cross-section of the ion column and sample stage in the vacuum chamber

37

ŝ

uum of  $< 10^{-7}$  torr. Milling patterns are created using start and end point files to ensure precise relative locations of the milled areas, as illustrated in Fig. 4.3(c). The distances between the milled strips are increased from 0.1 µm in steps of 0.1 µm from the left to right. The milled depth is controlled by the milling time. The milling starts from the left most strip, and proceeds one by one to the right. When the enhanced etch is used, iodine is



(a)



(b)



Fig. 4.3. Illustration of the FIB milling process (a-b) and milling pattern (c) created from a start and end point file. The shaded area are defined to be milled in (c).

introduced into the chamber through a small hollow needle after the iodine source is heated to 30°C. The introducing of iodine not only accelerates the milling rates, but also helps to remove the milled materials by forming volatile compounds that are pumped away by the vacuum system, preventing any material redeposition. In order to observe the significance of redeposition during FIB milling, the same pattern is also milled without introducing the enhanced etching agent. An additional strip is milled perpendicular to the standard pattern which overlaps the top part of the pattern for identification.

## 4.3 Results and Discussion

It is accepted that the materials milled from FIB micromachining would remain in the vicinity of where they are sputtered off rather than being swept away into the vacuum system. These redeposited materials are expected to adhere loosely and be mechanically unstable. However, to the author's knowledge, no research has been done to observe the significance of redeposition when only a small amount of material (such as the case in this study) is sputtered off from any type of milling target. The purpose of this experiment then was to see whether it is plausible to mill features using FIB without using an etch enhancing agent for reasons we shall discuss later in this section.

In fig. 4.4(a) we show the SEM image of the milled features without using enhanced etching. Redeposited materials are clearly seen to accumulate along the edges of the milled strips. These materials cannot be removed by additional milling without introducing modification to the substrate, even though the milling rate for the redeposited materials is expected to be significantly higher than that for the substrate. The top part of the SEM image in fig. 4.4(a) shows an area that received additional milling perpendicular to the strip pattern. It is seen that the surface is higher at the location where the materials are redeposited.

Fig. 4.4(b) shows the SEM image of the features milled using the enhanced etch process. Redeposition of materials is not observed. Instead, the substrate along the edge of



Fig. 4.4. SEM image of the features milled (a)without using enhanced etch and (b) using enhanced etch

40



Fig. 4.5. 3-D AFM image of the feature milled using enhanced etch .



Fig. 4.6. (a) Schematics of the wedge shape. The slope of the wedge is about  $3^{\circ}$ . (b) The shape of the feature expected from FIB milling with the presence of iodine if a milling depth of 100 Å is required and the center of the feature is not milled. The dotted box shows the ideal Ge bar if no tail effect of the FIB is involved.



Fig. 4.7. The calculated current density distribution for a FIB system at two different states of focus, one minimizing the rise distance of the beam across a knife edge and one optimizing the spatial resolution [4.20-4.21].

the strip patterns is milled away, leaving wedges in the place. The shape of these wedges has been further studied using AFM. A 3-dimensional AFM image of the feature is shown in Fig. 4.5. The trenches in the image are strips of material removed by the FIB process. The slope of the edges of these trenches is measured to be  $\sim 3^{\circ}$ , as shown in Fig. 4.6(a).

This results from the milling effect of the FIB tail. We show in Fig. 4.7 the calculation results of the current density of the FIB. The FIB tail is several orders of magnitude lower than that at the FWHM of the FIB [4.20-4.21]. Evidently, the milling efficiency of the FIB tail is enhanced more than that of the center of the beam with the presence of iodine. If a milling depth of 100 Å is required to remove a Ge thin film and a thickness of 100 Å is to be preserved at the center of the feature, FIB milling will only produce a pyramid of 100 Å high and 0.38  $\mu$ m long at the bottom (Fig. 4.6(b)) instead of the rectangular bar (Ge wire) for the ideal case.

Contamination from the enhanced etch is observed as droplets randomly distributed in the milling area, indicated by the arrows in Fig. 4.8. Energy Dispersive X-ray (EDX) analysis shows a slight trace of iodine in the droplets. The droplets are not soluble in isopropyl alcohol.



Fig. 4.8. SEM image of Ge wire array fabricated by FIB micromachining from a 50 Å Ge/Si sample using iodine as enhanced etch agent. The arrows point to the droplets randomly scattered over the milling area. EDX shows a slight amount of iodine in the droplet.



Fig. 4.9. XTEM image of (a) the wedge at one side of the Ge wire fabricated using a FIB dose of  $1.8 \times 10^{14}$  cm<sup>-2</sup> and (b) a sample received a FIB dose of  $1.12 \times 10^{14}$  cm<sup>-2</sup>. The Ge film thickness before FIB milling is 50 Å.

Cross-sectional transmission electron microscopy (XTEM) is used to help evaluate the quality of the Ge wires fabricated by FIB micromachining. The wedge shape of the edge of a Ge wire is shown in Fig. 4.9(a), from which we draw the following conclusions. Firstly, the image shows a ~ 300 Å thick Si layer beneath the Ge is amorphized in places where the Ge film is just completely removed. The height of the Ge layer above the amorphized Si is slightly higher than the Ge film which received no FIB milling. We believe this is a result of the effect of Ga ion implantation and resulting volume expansion occurred during Si amorphization. Secondly, the Ge film milled by the FIB is diffused, indicating the Ge atoms are "knock-on" displaced into the silicon substrate following collisions with the energetic Ga beam, even though some Ge atoms are removed by sputtering.

The threshold FIB dose for removal of 50 Å of Ge thin film is also investigated. In Fig. 4.9(a), the Ge film thickness is about 50 Å and the total FIB dose used is about  $1.8 \times 10^{14}$  cm<sup>-2</sup>. The Ge film as well as a thin layer of Si is observed removed. Fig. 4.9(b) shows a sample for which a FIB dose of  $1.12 \times 10^{14}$  cm<sup>-2</sup> is used. We see that the Ge film is not completely removed from the trench. Hence, the threshold dose for removal of 50 Å amorphous Ge lies between the two above mentioned values. Notice that in Fig. 4.9(b), significant amorphization of the Si substrate beneath the Ge film is induced before the Ge layer is completely removed.

## 4.4 Conclusions and Future Work

In conclusion, we have investigated the direct FIB micromachining technique to fabricate Ge nanowires. Without enhanced etch, significant redeposition of milled materials along the edge of the wires is observed even though the milling depth is extremely shallow (several hundred angstroms). Use of enhanced etch during the FIB milling avoids redeposition of the milled materials. However, several major problems still exist using enhanced etch FIB milling. First, milling effects resulting from the FIB tail produces wedge shaped Ge wire edges, limiting the possible minimum width capability for the Ge wire. Secondly, introducing iodine for the enhanced etch process causes the formation of droplets of unknown composition randomly scattered over the milling area. Corrosion resulting from the remains of the iodine on the silicon substrate has been observed by other research groups [4.23]. Lastly, FIB milling modifies the composition of the target through intermixing effects resulting from the implantation of the Ga ions.

Future work should focus on resolving the above mentioned problems. As an example, a sacrificial layer (such as PMMA or  $SiO_2$ ) with low FIB milling rates may be used on the Ge film to minimize the wedge effect. Chemical cleaning methods should be developed for removal of the contamination resulting from the enhanced etch agent.

#### 4.5 Reference

- [4.1] W. Escovitz, T. Fox, and R. Levi-Setti, "A key kev proton scanning microscope", Proc. of the Annual Meeting of the Electron Microscopy Society of America (EMSA), Vol. 32, 416(1974).
- [4.2] R. Levi-Setti, "proton scanning microscopy:feasibility and promise", Scanning Electron Microscopy, 125(1974).
- [4.3] W. Escovitz, T. Fox, and R. Levi-Setti, "metaphase chromosomes in the scanning proton microscope", Proc. of the Annual Meeting of the Electron Microscopy Society of America (EMSA), Vol. 33, 304(1975).
- [4.4] J. H. Orloff and L. W. Swanson, "study of a field ionization source for microprobe applications", J. Vac. Sci. Technol. Vol. 12, 1209(1975).
- [4.5] J. Orloff and L. W. Swanson, "a scanning ion microscope with a field ionization source", Scanning Electron Microscopy, Vol. 10, 57(1977).
- [4.6] J. Orloff and L. W. Swanson, "fine focus ion beams with field ionization", J. Vac. Sci. Technol. Vol. 15, 845(1978).
- [4.7] P. D. Prewett and G. L. R. Mair, Focused Ion beams and Liquid Metal Ion

Sources, Wiley, Chichester, 1991.

- [4.8] L. W. Swanson, "Use of the liquid metal ion source for focused beam applications", Applied Surface Science, vol.76-77, 80(1994).
- [4.9] J. Melngailis, "Focused ion beam technology and applications", J. Vac. Sci. Tech. B, vol.5, 469(1987).
- [4.10] Steckl, Corelli, J.C.; McDonald, J.F.; Jin, H.S.; Higuichi-Rusli, R., "Focused ion beam technology and applications", in *Emerging Technologies for In Situ Processing. Proceedings of the NATO Advanced Research Workshop*, D. J. Ehrlich, Van Tran Nguyen (editors), Dordrecht, Netherlands: Martinus Nijhoff, 179 of x+290(1988).
- [4.11] G. Ben Assayag, P. Sudraud, "Metal ion and cluster beams for microelectronic research and development: a review", Scanning Microscopy, vol.2, no.3, 1329(1988).
- [4.12] K. Gamo, S. Namba, "Recent advance of focused ion beam technology in maskless deposition and patterning", Nucl. Instrum. & Meth. in Phy. Res., Sec. B, vol.B59-B60, pt.1, 190(1991).
- [4.13] K. Nikawa, "Applications of focused ion beam technique to failure analysis of very large scale integrations: a review", J. Vac. Sci. Tech. B, vol. 9, no. 5, 2566(1991).
- [4.14] M. Komuro, "Novel applications of focused ion beam technology", in Extended Abstracts of the 1991 International Conference on Solid State Devices and Materials, Tokyo, Japan: Bus. Center Acad. Soc. Japan, 53 of xvi+770(1991).
- [4.15] J. Bond, S. Cazzani, "Electron and ion beams for measurement and repair of integrated circuits", *Elettronica Oggi*, no. 137, 71(1992).
- [4.16] J. Melngailis, "Focused ion beam lithography", Nucl. Instrum. & Meth. in Phy. Res., Sec. B, vol.B80-81, pt. 2, 1271(1993).
- [4.17] Y. Ochiai, R. J. Young, J. R. A. Cleaver, H. Ahmed, T. Baba, "Fabrication of

microstructures for quantum devices using focused ion beam gas-assisted etching", Microelectronic Engineering, vol.13, no.1-4, 399(1991).

- [4.18] G. J. Young, J. R. A. Cleaver, H. Ahmed, "Characteristics of gas-assisted focused ion beam etching", J. Vac. Sci. Tech. B, vol.11, no.2, 234(1993).
- [4.19] J. D. Casey, A. F. Doyle, R. G. Lee, D. K. Stewart, H. Zimmermann, "Gas-assisted etching with focused ion beam technology", *Microelectronic Engineering*, vol.24, no.1-4, 43(1994).
- [4.20] P. Marriott, Appl. Phys. A. vol. 44, 329(1987).
- [4.21] J. Orloff, "High-resolution focused ion beams", Rev. Sci. Instrum. vol. 64 no. 5, 1105(1993).
- [4.22] W. K. Chu, H. Muller, and J. W. Myer, and T. W. Sigmon, "Residual disorder in Si from oxygen recoils in annealed "through-oxide" arsenic implants", *Appl. Phys. Lett.*, 25(5), 297(1974).
- [4.22] J. Puretz, private communication.

## Chapter 5

# GE NANOWIRE FABRICATION USING E-BEAM LITHOGRAPHY

### 5.1 Introduction

Electron beam (e-beam) lithography is the process of forming patterns by using a focused electron beam. Such beams can be readily scanned and accurately positioned on the substrate for exposing radiation-sensitive material (e-beam resist). Following early demonstrations of high-resolution patterning using focused electron beams [5.1-5.2], e-beam lithography has become a popular technique for delineating devices and structures on the scale of tens of nanometers [5.3-5.9].

E-beam lithography has several advantages over standard optical lithography. Firstly, it provides the ability to produce features without the use of a mask under direct control of a computer. Secondly, the equivalent wavelength of an electron beams is on the order of 0.2 to 0.5 Å. Therefore, diffraction effects, which limit resolution in optical lithography, are avoided. The resolution of e-beam lithography is limited by the so-called "proximity effect" which results when the electrons are forward scattered in the resist and back scattered from the substrate, resulting in exposure of the resist outside the desired areas. The proximity effect may be reduced by using thinner resists and higher energy electron beams to reduced forward scattering. However, the resist sensitivity drops as the primary beam energy increases. The contribution to the proximity effect from the back scattered electrons also increases with the atomic number of the substrate. As a result, the minimum line width obtainable on a Ge/Si substrate will be larger than that for a bare Si substrate for the same conditions.

Critical to nanostructure patterning is the properties of the selected resist material

used for processing. Two types of electron resists are available: negative and positive electron resists. Negative electron resists have excellent sensitivity, but relatively low resolutions. This is due to exposed resist swelling during development for removal of unexposed resists. For this reason, positive electron resists are more frequently used for nanostructure patterning. Continuing efforts are underway in the research community to develop high sensitivity, high resolution electron resists with good adhesive and mechanical properties [5.10-5.12]. For our experiment, we select the most widely used positive e-beam resist: poly(methylmethacrylate), or PMMA in this work.

In this chapter, we describe the lift-off process used to fabricate Ge nanowires using e-beam lithography. Efforts to obtain the threshold e-beam dose for exposure of PMMA at a certain film thickness for nano-patterning are described. Field emission SEM, AFM, and cross-sectional TEM are used in the characterization of these nanowires. In order to prepare the TEM sample of the nanowires, an advanced technique using multiple protection layers and FIB marking and thinning is developed.

### 5.2 Experiment

The process used in our experiment is illustrated in Fig. 5.1. Details of the experiment are as follows: first, an n-type (100) Si substrate is given a RCA clean followed by a 30 s 1:50 HF:H<sub>2</sub>O dip. The sample is then spin coated with 1.5% PMMA dissolved in chlorobenzene. After being baked at 160°C for 8 hours, a thickness of about 150 nm of PMMA is obtained. The sample is then exposed in line patterns using a custom designed e-beam lithography system [5.13]. The scanned lines are located in an area of 47 x 80  $\mu$ m<sup>2</sup>. Various electron beam dosages, ranging from 0.9 to 1.4 nC/cm, are used for various sets of lines in the same pattern to determine beam dose effects on the quality of the Ge wires. The sets are separated by strip patterns for which the area scan is used. Next, the sample is developed for 80 s in a 1:3 MIBK:IPA solution at 25°C followed by a 15 s DI water rinse. Immediately prior to loading the sample into an electron beam evaporator, it



(a) A thin PMMA film is patterned by an electron beam.



(b) Exposed PMMA is developed in MIBK:IPA solution.



(c) A thin film of Ge is deposited using e-beam evaporation.



(d) Ge wires are formed after a lift-off process in acetone.

Fig. 5.1 Illustration of the lift-off process to fabricate Ge nanowires using e-beam lithography. is exposed to HF vapor for 10 s to passivate the surface. A thin film of amorphous Ge, typically 50-100 Å, is then deposited at room temperature under a vacuum  $< 10^{-7}$  torr, similar to the process described in Chapter 4. The Ge wires are formed during the lift-off process in acetone.

# 5.3 Characterization and XTEM Sample Preparation

Since the Ge wires are extremely thin and narrow, field emission SEM is require ed in order to view the wires effectively. After examination using the field emission SEM and AFM, the sample is ready for cross-sectional TEM specimen preparation.

There are a number of methods for preparation of cross-sectional TEM specimens [5.14-5.16]. The conventional specimen preparation technique widely used consists of mechanical polishing, dimpling and ion milling. This method works very well with homogeneous, large samples. However, it is very difficult to apply to micron-scale structures. A common practice is the use of masks for selective ion beam milling. This method is cumbersome, and for micron-scale devices in a small area, does not ensure that the area of interest will be thinned to a uniform thickness for examination. In this work, the Ge wires are small and located in a specific area smaller than 100  $\mu$ m on a side, It is physically very difficult to preselect this area in conventional specimen preparation techniques even using the above described masking technology. We therefore decided to use a FIB assisted milling technique for this process.

Cross-sectional TEM sample preparation using FIB assisted milling is a newly developed technology [5.17-5.26]. For purpose of convenience, different researchers use slightly different approaches. The general approach, shown in Fig. 5.2, is to cut the cross-section of the sample using a slow speed diamond saw as the first step, and make sure the area of interest is in this sample. Next the sample is mechanically polished to a thickness of up to 10  $\mu$ m. The thinner the sample after polishing, the less time is required for FIB milling. However, a certain thickness is often required to ensure that the area of interest is



(b)





Fig. 5.2. Illustration of the cross-sectional TEM sample preparation process using FIB micromachining. (a) Mechanically saw a slice from the wafer containing the area of interest. (b) Mechanically polish this slice to > 10 mm. (c) Mount the slice on a modified TEM grid and mill two trenches, leaving a strip of materials which contains the feature of interest. The grid is then mounted to a support.

safe from damage. Once the desired thickness is reached, the sample is trimmed to a length that fits on the mount. Next, the sample is mounted to a modified grid which prevents the grid from blocking the ion beam during ion milling later in the process. If the sample is too long for the grid, the side(s) of the sample is trimmed away before mounting. The grid is then mounted on a support, which holds the sample straight. The support used in this work is a simple vise, as shown in Fig. 5.2(c). This vise can hold several samples at one time. After the sample is transported to the FIB station, two trenches are milled in it, leaving a narrow strip (or foil) of material surviving between them which forms the TEM section, within which the area of interest is contained. The milling is done with several different FIB currents, which gives different milling rates and beam sizes. To minimize the milling time, our milling starts at the largest available beam current, often as high as 13, 000 pA, depending on the condition of the ion source. The final milling is performed using a beam current of 16 pA, to obtain lattice resolution grade samples. Since the beam tail of the FIB may mill away features at the surface of the sample, FIB assisted in situ chemical vapor deposition (CVD) of a metal layer, such as the Pt shown in Fig. 5.3(a), is used in this work to serve as a protection layer. This is often performed on the area of interest before any milling [5.23]. Since a strong contrast exists between the Pt layer and the Si substrate for the focused ion microscopy image obtainable on the FIB machine, we use this layer of material as an indication wether the foil reaches the desired thickness. In situ observation, often in an area on the foil which does not contain the feature of interest, is performed from time to time using focused ion microscopy by tilting the sample in the FIB chamber when using the lowest beam current. A rule of thumb is that the foil is typically thin enough to obtain lattice imaging once the Pt is almost milled away at a beam current of 16 pA for a Pt layer thickness of 1  $\mu$ m thickness when the milling was started. Fig. 5.3(b) shows a FIB made TEM sample with a small amount of remaining Pt on top of the foil between the two trenches.

Due to the small physical dimensions of the Ge wire array, a FIB marking tech-



(a)



(b)

Fig. 5.3. Scanning ion image of (a) the Pt bar deposited by FIB assisted CVD on the area of interest and (b) the foil made by milling two trenches and leaving a strip of materials in between. The Pt is visible in the image because of its distinguished contrast with the Si substrate. Traces of Pt on the top of the foil are used as an indication of the foil thickness in the milling process.



nique [5.27-5.28] is used for location of the feature throughout the process. Immediately before the sample is subjected to XTEM sample preparation, it is transferred to the FIB workstation and two "T" shaped marks are made at the sides of the Ge wire array, as shown in Fig. 5.4. The lowest FIB current available, e.g. 6 pA, is used for locating the fea-

ture and care is used to avoid removing the Ge nanowires when using the FIB viewing scan.

The above process works well except that Ge or Si inter-diffuses into the Pt protection metal layer deposited by FIB assisted CVD. This problem becomes more serious due to localized heating from the ion beam when applied to an extremely thin foil of material(s). A diffusion barrier is needed in between the metal layer and the Ge/Si sample. Therefore, we developed a multilayer scheme to address this problem. As shown in Fig. 5.5, a thin carbon layer, about 100 Å thick, is thermally evaporated onto the sample, followed by a ~1200 Å layer of Au-Pd alloy, deposited using plasma sputtering. The final Pt



Fig. 5.5. Multilayer protection scheme to protect the surface of the substrate in the TEM, sample preparation process using FIB milling, particularly when the feature of interest is on the surface. (a) A thin layer of C( $\sim$  100 Å thick) is thermally evaporated on the surface. (b) Au-Pd alloy ( $\sim$  1200 Å thick) is deposited using plasma sputtering.

layer is deposited on the top of this multilayer structure. The carbon layer serves as the diffusion barrier between the metal layer and the substrate. This layer cannot be replaced by  $SiO_2$ , which can be easily deposited by spin-on-glass. Fig. 5.6(a) shows a TEM sample prepared using spin-on-glass as the barrier layer. The surface of the Si substrate becomes amorphous in the foil, while the surface beneath a particle is protected and does not. We



Fig. 5.6. Cross-sectional TEM image of a sample prepared using (a) spin-on-glass or (b)a thin film of carbon (~ 100 Å thick) as the diffusion barrier. The surface of the Si substrate of both samples becomes amorphous, indicating neither scheme is ideal to protect an area of interest at the surface. The surface of the Si substrate underneath a particle is protected in image (a).

believe this amorphous layer is formed by the inter-diffusion between  $SiO_2$  and Si caused by localized heating from the ion beam because the  $SiO_2$  has a low thermal conductivity.



Fig. 5.7. Lattice image of a sample prepared using FIB milling with multilayer protection scheme. The foil made by this technique generally is believe to be wedge shaped, i.e. thinner at the top of the foil, thicker at the bottom. However, areas of lattice imaging quality in the foil can extend more than 1  $\mu$ m from the top of the foil, as in the case for this sample. The image is taken along the [110] orientation.

The Au-Pd layer is necessary since the carbon layer is too thin to block the diffusion of Pt into the substrate, as shown in Fig. 5.6(b). The diffusion of Pt probably begins during its deposition by FIB assisted CVD since Pt atoms carry along the energy of the FIB. The thickness of the carbon layer is limited by the amount of carbon which can be thermally evaporated each run by our particular thermal evaporator. Note the FIB marks help to locate the area of interest after the Au-Pd deposition. A high quality XTEM sample is obtained in 5 to 6 hours using this multilayer scheme. Fig. 5.7. is a high resolution TEM image with lattice resolution obtained from a sample using the above process. It is believed that the foils made from FIB milling is wedge shaped, i.e. thinner at the top of the



Fig. 5.8. Field Emission SEM image of three groups of Ge wires processed with e-beam dosages of 1.0, 0.95, 0.9 nC/cm from left to right respectively. The wires are not continuous when the e-beam dose used is blow 0.95 nC/cm.





foil and thicker at the bottom. However, with this particular sample, lattice resolution images are obtainable from the entire foil for more than 1  $\mu$ m along its depth. In addition, all features of interest are preserved in the foil for observation. This process is universally applicable to any localized surface structures.

## 5.4 Results and Discussion

Figure 5.8 shows a field emission SEM image of three groups of fine Ge wire structures processed using e-beam doses of 1.0, 0.95, 0.9 nC/cm from left to right, respectively. The three groups of wires are separated by Ge wires with broad width. It is understandable that a low dose is preferred because the lower the dose, the less the beam broadening effect and therefore, the narrower the wires would be. However, as seen in the figure, for the lowest e-beam dose, the PMMA does not completely develop and the lift-


off process results in noncontinuous wires. The threshold dose required to maintain continuous wires is about 1.0 nC/cm. As shown in Fig. 5.9, the wires are seen to be fairly straight. The width of the fine wires in the most left group in Fig. 5.8 is measured to be 57 nm. These wires exhibit a certain degree of roughness which is typical in features defined by lift-off processes. The AFM image (Fig. 5.10) better depicts the topology of the wires.

The thickness of the wires is measured to be 6 nm from the cross-sectional TEM image (Fig. 5.11). The TEM image shows that the wire has knife edges. We believe this is formed by oxidation of both Ge and Si after the sample is exposed to the air. This is in agreement with the fact that the silicon surface beneath the Ge wire is slightly higher than the rest of the surface. The thin transparent layer (5 Å), believed to be oxide, is visible beneath the Ge wire in the image. There are several possible reasons for the formation of this layer: (i) the native oxide on the Si substrate is not completely removed before e-beam



Fig. 5.11. Cross-sectional TEM image of a Ge wire on a Si substrate formed after a lift-off process using e-beam lithography.

evaporation or develops during the course of the e-beam evaporation; (ii) it is formed by oxidation during the FIB assisted specimen preparation. There is evidence showing the oxidation effect from the FIB milling. The 25 keV focused ions impinge on the side of the foil generating heat during the milling process. In some specimens prepared by FIB milling, the heating can produce an oxide layer of hundreds of nanometers in thickness on the surface of the Si. However, this oxide does not necessarily present a problem for the PLIE process because of its relative insensitivity to a very thin oxide layer [5.29], even if this thin layer is between the Ge wire and the Si substrate.

#### 5.5 Conclusion

In conclusion, Ge wires with cross sections as small as 6 by 57 nm are fabricated on silicon substrates using a lift-off process combining e-beam lithography/evaporation. Line scanning is used for the e-beam exposure of the PMMA thin film. The e-beam dosage used for the line scans needs to be larger than 1.0 nC/cm for the formation of continuous wires. Lower e-beam dosage results in discontinuous wires. An advanced crosssectional TEM specimen preparation procedure is developed to examine the wires. The procedure is applicable for any cross-sectional sample preparation using FIB micromachining for localized surface structures.

#### 5.6 Reference

- [5.1] G. Mollenstedt and R. Speidel, Phys. Bl. 16, 192(1960).
- [5.2] A. N. Broers, Proceedings of the First International Conference on Electron and Ion Beam Science and Technology, edited by R. Bakish (Wiley, New York, 1965), 191(1965).
- [5.3] P. M. Mankiewich, R.E. Howard, L. D. Jackel, W. J. Skocpol, and D. M. Tennent,J. Vac. Sci. Technol. B4, 380(1986).
- [5.4] C. G. Smith, H. Ahmed, and M. N. Wybourne, J. Vac. Sci. Technol. B5, 314(1987).

- [5.5] T. A. Fulton, and G. L. Dolan, Phys. Rev. Lett. Vol 59, 109(1987).
- [5.6] D. R. Allee, A. N. Broers, and R. F. W. Pease, "Limits of nano-gate fabrication", Proc. of the IEEE, Vol. 79, 1093(1991).
- [5.7] H. Ahmed, "Nanostructure fabrication.", Proc. of the IEEE, Vol. 79, 1140(1991).
- [5.8] C. R. K.Marrian, E. A. Dobisz, and M. C. Peckerar, "Nanostructure patterning", Proc. of the IEEE, Vol. 79, 1149(1991).
- [5.9] T. Nishida, M. Notomi, R. Iga, and T. Tamamura, "Quantum wire fabrication by E-beam lithography using high-resolution and high-sensitivity E-beam resist ZEP-520", Jap. J. of Appl. Phys., Part 1, Vol. 31, No. 12B, 4508(1992).
- [5.10] E. Roberts, "Recent Developments in Electron Resists", Solid State Technol., 135(1984).
- [5.11] E. Reichmanis, L. F. Thompson, "Polymer materials for microlithography", Annual review of materials science. Vol. 17, edited by R. A. Huggins, J. A. Giordmaine, J. B. Wachtman, Jr., 235(1987).
- [5.12] S. Affrossman, M. Baschee, D. R. Brambley, F. L. Chow, J. M. G. Cowie, C. Dix,
  L. G. Griffiths, P. Hendy, M. E. Jones, R. G. Jones, P. Miller Tate, M. Moshin, Pet,
  "Development of resists for electron beam lithography", UK IT 88 Conference
  publication, Swansea, UK, July, 1988, UK: Inf. Eng. Directorate, 563(1988).
- [5.13] J.C. Nabity and M.N. Wybourne, "A versatile pattern generator for high-resolution electron-beam lithography", *Rev. Sci. Instrum.*, Vol. 60, No. 1, 27(1989).
- [5.14] L. M. Gignac and A. L. Edel, "Sample preparation methods for TEM analysis of semiconductor devices", presented at 17th Intern. Sym. for Testing & Failure Analysis, Los Angles, California, USA, Nov., 1991.
- [5.15] R. Anderson, S. Klepeis, J. Benedict, W. G. Vandygrift, M. Orndorff, "Recent developments in the preparation of semiconductor device materials for the transmission electron microscope", in Microscopy of Semiconducting Materials 1989. Proc. of the Royal Microscopical Soc. Conf. Edited by A. G. Cullis, J. L. Hutchin-

son, Bristol, UK: IOP, 1989. 491-500 of xvi+831(1989).

- [5.16] H. Cerva, H. Oppolzer, "Electron microscopy for thin film investigations", Vakuum-Technik, Vol. 38, No. 5-6, 143(1989).
- [5.17] Y. Hata, R. Etoh, H. Yamashita, S. Fujii, Y. Harada, "Microstructure analysis technique of specific area by transmission electron microscopy", *IEICE Transactions on Electronics*, Vol. E77-C, No. 4, 590(1994).
- [5.18] C. Vieu, A. Pepin, G. Ben Assayag, J. Gierak, F. -R, Ladan, "Localized thinning of semiconductor nanostructures for cross-sectional transmission electron microscopy", in *Microscopy of Semiconducting Materials 1993*. Proceedings of the Royal Microscopical Society Conference. Edited by A.G. Cullis, A. E. Staton-Bevan, J. L. Hutchison, Bristol, UK: IOP, 1993. 385 of xviii+788(1993).
- [5.19] M. Tarutani, Y. Takai, R. Shimizu, K. Uda, H. Takahashi, "Development of a focused ion beam apparatus for preparing cross-sectional transmission electron microscope specimens", *Technology Reports of the Osaka University*, Vol.43, No. 2142-2162, 167(1993).
- [5.20] G. Ben Assayag, C. Vieu, J. Gierak, H. Chaabane, A. Pepin, P. Henoc, "Localized thinning of GaAs/GaAlAs nanostructures by a combined scanning electron micrograph/focus ion beam system for high-quality cross-sectional transmission electron microscopy samples", J. Vac. Sci. Tech. B, Vol. 10, No. 3, 531(1993).
- [5.21] A. Yamaguchi, M. Shibata, T. Hashinaga, "Transmission electron microscopy specimen preparation technique using focused ion beam fabrication: application to GaAs metal-semiconductor field effect transistors", J. Vac. Sci. Tech. B, Vol. 11, No. 6, 2016(1993).
- [5.22] K. Nakajima, S. Sudo, M. Yakushiji, T. Ishii, S. Aoki, "Layer structure evaluation of multilayer X-ray mirror by combination of focused ion beam etching and transmission electron microscopy", J. Vac. Sci. Tech. B, Vol. 11, No. 6, 2127(1993).
- [5.23] D. P. Basile, R. Boylan, B. Baker, K. Hayes and D. Soza, "FIBXTEM-focussed

ion beam milling for TEM sample preparation", in Specimen Preparation for Transmission Electron Microscopy of Materials - III Symposium. ed. R. Anderson, B. Tracy, J. Bravman, (Mater. Res. Soc, 1992.) p. 23-41.

- [5.24] R. J. Young, E. C. G. Kirk, D. A. Williams, and H. Ahmed, "Fabrication of planar and cross-sectional TEM Specimens using a focused ion beam", in Specimen Preparation for Transmission Electron Microscopy of Materials - II, edited by R. Anderson (Mater. Soc. Proc. 199, Can Francisco, California, 1990), 205(1990).
- [5.25] K. -H. Park, in Specimen Preparation for Transmission Electron Microscopy of Materials - II, edited by R. Anderson (Mater. Soc. Proc. 199, Can Francisco, California, 1990), 271(1990).
- [5.26] E. C. G. Kirk, D. A. Williams, and H. Ahmed, in Proc. of the 6th Microscopy of Semiconducting Mater. Conf., Inst. Phys. Conf. Cer. No. 100: Section 7, 501(1986).
- [5.27] C. E. Sanborn, S. A. Myers, "Precision TEM sample preparation using focused ion beam marking strategies", in Specimen Preparation for Transmission Electron Microscopy of Materials - III Symposium, edited by R. Anderson, B. Tracy, J. Bravman, Pittsburgh, PA, USA: Mater. Res. Soc, 239(1992).
- [5.28] N. Nishio, K. Shiotani, M. Kitakata, M. Mikoshiba, "FIB bit marking technique for XTEM analysis of megabit memory failures", in *Extended Abstracts of the* 22nd (1990 International) Conference on Solid State Devices and Materials, Tokyo, Japan, 1169 of xiii+1247(1990).
- [5.29] J. R. Abelson, T. W. Sigmon, Ki Bum Kim and K. H. Weiner, "Epitaxial Ge<sub>x</sub>Si<sub>1-x</sub>/ Si(100) structures produced by pulsed laser mixing of evaporated Ge on Si(100) substrates", Appl. Phys. Lett. 52(3), 230(1988).

#### Chapter 6

# FABRICATION OF SIGE NANOWIRES USING PULSED LASER INDUCED EPITAXY

#### 6.1 Introduction

In this chapter, we address the primary goal of this work to demonstrate direct formation of SiGe nanowires in Si using the pulsed laser induced epitaxy process. In the previous chapter, we described the successful development of a process to obtain Ge nanowires on a Si substrate and the corresponding characterization techniques. Specifically, Ge nanowires as narrow as 57 nm are demonstrated using a lift-off process with ebeam lithography. The thickness of Ge wires is readily controlled by e-beam evaporation, which has an accuracy of  $\pm$  5 Å [6.1]. This provides the ability for obtaining ultra-narrow SiGe wires by directly incorporating the Ge into the Si substrate with the PLIE process. Our advanced TEM specimen preparation procedure, described in Chapter 5, allows high resolution cross-sectional transmission electron microscopy analysis of the SiGe nanowires with a high sample preparation yield.

One concern we have neglected to date is the side diffusion effect occurring in the PLIE process as configured in our experiments. In the PLIE process, both Ge and the Si substrate are melted. The energetic laser pulse causes a rapid melt and regrowth process in the new surface region of the samples. The laser energy is absorbed very close to the surface as the absorption depth in Si for the 308 nm laser beam is about 7 nm and roughly 10 nm in Ge [6.2, 6.3]. Recrystallization occurs from the single-crystalline substrate. During the rapid melt process, the Si and Ge intermix to form localized alloy regions. When masks are used in the process, such as discussed in reference 6.1, where 3-µm wide, 200 nm deep SiGe wells are formed (Fig. 6.1), the Si substrate beneath the mask is not melted

limiting Ge diffusion to the un-masked area. In our work, areas other than those beneath the Ge nanowires described in the previous chapter are also subject to laser irradiation, and thus melt, as shown in Fig. 6.2, resulting in the possibility of side diffusion. This should not be significant, since the melt duration is on the order of tens of nanoseconds. However, the side diffusion may be enhanced by the fact that Ge is melted first since the melting starts from the surface of the sample and the Ge has a lower melting temperature than Si.

With this in mind, we pursue fabrication of the SiGe nanowires by two methods: In the first method we apply the laser beam directly onto the Ge nanowires/Si sample. This is a straight forward and simple process without additional steps. In the second method, we



Fig. 6.1. Cross-sectional TEM image of SiGe wells formed in a PLIE process using an Al/SiO2 mask. No side diffusion is observed since the substrate area beneath the mask is not melted in the PLIE process. (Ref. 6.1)



Fig. 6.2. Illustration of a possible melting and diffusion process in a PLIE process by directly applying the laser beam on the Ge nanowire on Si.

deposit a thin LTO capping layer on top of the Ge nanowires/Si sample. The thin LTO layer is essentially transparent to the 308 nm XeCl laser beam and does not hinder the melting process. However, the LTO layer is expected to prevent the Ge nanowires from ablation/evaporation when they are melted.

In the remainder of this chapter we present the details of our efforts to fabricate SiGe nanowires in Si. Section 5.2 describes the method without using the LTO capping layer. Interesting surface evolution observed after the PLIE process other than lateral diffusion is discussed based on the results obtained using SEM, Nomarski microscopy, crosssectional TEM, atomic force microscopy, and Auger electron spectroscopy. Section 5.3 describes results obtained when using the LTO capping layer and are compared with those obtained from the first method. The final section summarizes our results.

## 6.2 Formation of SiGe Wires Without LTO Capping Layer

Details of this experiment are as follows: n-type (100) Si substrates are patterned



Fig. 6.3. Scanning electron micrograph of a Ge wire array on Si before laser irradiation.



Fig. 6.4. Scanning electron micrograph of the Ge wire array shown in fig.6.3 after laser irradiation.

with Ge wire arrays using the lift-off process and e-beam lithography described in the previous chapter. The Ge wires are ~ 60 nm in width and ~ 6 nm in thickness with spacings of either 0.5  $\mu$ m or 1  $\mu$ m for different samples. As shown earlier in the SEM image (Fig. 1a), the six arrays of wires are divided by broader Ge wires of widths of 0.2  $\mu$ m, 0.6  $\mu$ m, 0.8  $\mu$ m, 1  $\mu$ m and 2  $\mu$ m. Lower e-beam dosages are used for these broader lines resulting in a partial lift-off of Ge films in these wires, as seen in the image. The PLIE process is carried out in the gas cell through a quartz window. Immediately prior to placing the samples into the gas cell, the samples are rinsed in DI water for 20 s followed by 15 s HF vapor exposure to passivate the surface. The gas cell is pumped out to a pressure of less than  $1 \ge 10^{-7}$ Torr and then backfilled with Ar gas to ~ 500 Torr. The samples are then irradiated through the quartz window using the spatially homogenized 308 nm XeCl excimer laser beam at an energy fluence of ~  $0.82 \text{ J/cm}^2$ . The duration of the laser pulse is 27 ns. The homogenized beam is a square of 3 mm on a side. The applied laser beam covers the entire wire pattern as well as some of the surrounding area. The melting and resolidification process is monitored using a HeNe laser beam, which is used in situ to probe the sample surface exposed to the laser beam to record reflectance changes during the process. The surface melt duration at this energy fluence for the bare Si surface is found from the reflectance data to be ~ 40 ns.

## Characterization and Discussion

Fig. 6.4 shows an SEM image of the same sample shown in Fig. 6.3 following application of a single laser pulse. Ripples are formed on the surface in regions where the Ge wire arrays are located. From this image, the original Ge wire locations are viewed as plateaus above the surface. This seems to conform with the fact that the original Ge wires are sitting on the surface of the substrate and their original shape is partially preserved after intermixing with the Si. However, cross-sectional TEM analysis reveals that this is *not* the case.

In fig. 6.5(a) we show a TEM micrograph which illustrates the topology of the sur-



(a)



**(**b)

Fig. 6.5. Cross-sectional TEM image following one laser irradiation of (a) Ge wires/Si with 0.5  $\mu$ m spacing (b) Ge wires/Si with 1.0  $\mu$ m spacing.

face of a sample with a spacing of  $0.5 \,\mu$ m between the wires. The two large ripples seen in the image result from the original 1- $\mu$ m wide Ge wire. To the left of these two large ripples can be seen several small ripples from the 60 nm wide Ge wire array. Apparently, the original Ge wire locations have become troughs rather than plateau after laser irradiation. The height of the ripples formed by the 60-nm wide Ge wires with 1  $\mu$ m spacing (Fig. 6.5(b)), as measured from TEM, is about 22 nm, much higher than the original height (6 nm) of the Ge wires deposited on the Si surface. The topology of samples irradiated with different numbers of laser shots are also examined by AFM. As can be seen in Fig. 6.6, the ripple height decreases with increasing number of laser shots. Fig. 6.7 is a Nomarski inter-







Fig, 6.7. Nomarski micrograph for a wire sample after laser irradiation. Trenches can be seen surrounded by ripples.

ference image which shows more detail of the ripple features formed on the surface. The troughs formed by the Ge wires surrounded by ripples are obvious. The roughness on the surface of the broader wires results from the partial absence of the Ge film. For the 60-nm wide Ge wire array, quasi-sinuous ripples are formed. However, for the broader Ge wires, the original Ge film height is still partially preserved. This can be seen in the surface topology formed by the 1- $\mu$ m wide wire in Fig. 6.5(a). The SEM image in Fig. 6.3 has probably picked up more composition related contrast than topology contrast.

The reason for ripple formation is not fully understood at this time. One possible explanation follows. As illustrated in Fig. 6.8, during the 40 ns melt time of the PLIE process, Ge and Si inter-diffuse forming SiGe alloy. At the same time, particularly at the beginning of the melting process the Ge wires move downward because of mass and gravitational effects interacting with the Si surface resulting in the formation of the ripples along the sides of the wires. This is similar to the case of ripple formation on a water surface when a drop of ink is put in. However, in the PLIE process, ripples are frozen in rather than disappearing as they are formed. the force of gravity, aided by the Brownian motion of the atoms, also aids the diffusion of Ge into the Si to form a compositionally uniform alloy of SiGe, as shown in Fig. 2.5. A factor which counteracts the ripple formation process is surface tension of the liquid Si. This results in the ripples formed from the 1-µm spacing 60-nm wide Ge wire array in Fig. 6.5(b) being larger than those formed in the 0.5-µm spacing 60-nm Ge wire array of Fig. 6.5(a). Similar surface evolution is observed for the PLIE process in fabricating SiGe nanowires when using a thin LTO capping layer on top of Ge wires/Si structures. We discuss this topic further in the next sec-



Fig. 6.8. Illustration of the ripple formation process





tion.

Since the SiGe/Si mass contrast is not observed in the TEM image, we can not determine the boundary of the Ge diffusion. We use Auger Electron Spectroscopy (AES) to determine the lateral diffusion of the Ge. In order to retain reasonable resolution without losing intensity of Auger electrons, a beam current of 5 nA is used in the experiment. The beam energy used is chosen to be 1134 eV. Figure 6.9 shows Ge LMM Auger electron signals from three SiGe wires formed from 57-nm wide Ge wires and one formed from a 2-µm wide Ge wire. It is seen that all wires broaden in width. The intensity of the Auger signal is much lower for the three smaller wires than that for the 2-µm one, an indication of lower Ge concentration in the small wires. All three smaller SiGe wires consistently have a full width half maximum (FWHM) of about 300 nm. This indicates that significant lateral diffusion of Ge is induced in the 40-ns PLIE process. In the next section, we demonstrate how this lateral diffusion may be tremendously reduced simply by using a thin LTO capping layer on top of Ge nanowires/Si structure during the PLIE process.

## 6.3 Formation of SiGe Wires Using an LTO Capping Layer

In these experiments, experimental procedures similar to those described in section 6.2 are used with the exception of an additional LTO capping layer deposition step immediately before the PLIE process. A Ge nanowire samples is first degreased in hot TCE, acetone and methanol for 5 min each, followed by DI water rinse for 3 min. The sample is then dipped in a 2% HF:H<sub>2</sub>O solution for 15 s. After the sample is rinsed in DI water and dried in nitrogen, it is loaded into an LPCVD furnace for LTO deposition. The LTO is deposited to a target thickness of ~ 12 nm at a sample temperature of ~  $300^{\circ}$ C at a pressure of 225 torr. The sample is then transferred to the PLIE system. The PLIE process is carried out in air. The HeNe monitor laser beam is used in an area free of Ge wires for *in-situ* monitoring of the Si melt time. Figure 6.10 shows an example of the melt pulse and the laser pulse obtained for this experiment. The FWHM of the melt pulse is ~ 26 ns. This



Fig. 6.10. The melt pulse from an LTO covered Si substrate when a laser pulse (LLNL laser) is applied. The LTO thickness is ~ 15 nm. The FWHM of the melt pulse is 26 ns.



Fig. 6.11. SEM image of the SiGe wires with a thin LTO capping layer after laser irradiation.

implies an effective laser energy fluence of ~ 0.69 J/cm2 is used on the Si surface according to simulation results described in Chapter 2. Notice that the LTO layer reflects a small fraction of the laser beam [6.4].

Fig. 6.11 shows an SEM image of the Ge nanowire/Si sample after laser irradiation. The sample is still covered with the LTO capping layer as we shall see later in the TEM analysis. The 20 KeV electron beam in SEM Can easily penetrate this thin layer

(a)

(b)





Fig. 6.12.(a) Cross-sectional TEM image of a SiGe wire formed by PLIE using LTO as a capping layer. (b) illustration of the SiGe wire interface and its relative location with respect to the original Ge wire.

with little effect on the SEM image. It can be seen that the wires retain their original continuous nature after laser irradiation. However, a slight contrast variation along the wires is observable. The black particles in the image are possibly dust or contamination from the LTO deposition.

Fig. 6.12(a) is a cross-sectional TEM image of a SiGe wire. A slight contrast is seen between the SiGe region and the rest of the substrate. The interface of the SiGe/Si is spherical, as illustrated in Fig. 6.12(b). The width and depth of the SiGe wire in the Si substrate are estimated to be about 95 nm and 25 nm respectively. Instead of inter-diffusion of the Si and Ge as expected, Ge diffuses through the thin native oxide layer between the Ge layer and the Si substrate, mixes with the Si substrate forming the SiGe alloy. The thin native oxide layer is retained after processing and is observable in the TEM image. The mechanism behind the phenomena is unclear at this time. Possible reasons are: (i) the molten Ge produced in the process carries the momentum from the high flux of photons of laser beam, and has a tendency to move towards the substrate; (ii) the rapid melting of the Ge and substrate produces an electric field [6.5], which pulls the Ge into the substrate and mixes with the Si. In addition, the native oxide layer beneath the Ge is thin and porous, allowing the Ge to diffuse through. The diffusion of materials through SiO<sub>2</sub> layers are also observed in the metal gate stack Al/SiO<sub>2</sub>/Si system, where Si is observed to move through the SiO<sub>2</sub> layer segregating above Al after 12 hours annealing at 400°C [6.6].

High resolution TEM is performed on the samples. No defects are observed in the SiGe wire regions. The original Ge wire location is still seen in the image. Its slightly dark contrast when compared to the LTO is thought to be caused by Ge left behind in the LTO layer by incomplete diffusion. In some other locations, Ge is seen to have completely diffused into the substrate. Further experiments are needed to fully understand the cause of the incomplete diffusion. The Ge wire width is measured to be ~ 60 nm.

Comparing this process with the results obtained without using LTO capping layer, described in section 6.2, we see that the side diffusion of Ge is greatly reduced. However,





the side diffusion appears to be asymmetrical. We believe this is the result of the non-uniformity, including thickness and density, of the LTO capping layer deposited on top of the Ge wire. The variation in thickness and density of the LTO layer produces a local variation in the transmission of the laser beam through the  $SiO_2$  layer, which in turn produces a variation in the melt duration of the substrate. In figure 6.12, a porous region is seen in the image above the right portion of the SiGe wire. Fig. 6.13 is another cross-sectional TEM image which shows this non-uniformity effect. The LTO on the Ge wire is not uniform and is also thicker than the LTO deposited on the rest of the Si substrate. The maximum diffusion depth of the Ge into the substrate is smaller and the diffusion on the right side portion is more significant than on the left side. The Ge at the right side of the Ge wire completely diffuses into the Si substrate while some Ge remains on the left. Cleanliness of the Ge



Fig. 6.14. (a) SEM and (b) XTEM image of a contaminated Ge nanowires/Si sample after laser irradiation. A highly defected layer is formed at the surface of the substrate.

(a)

wire/Si surface is believed to be important for high quality LTO deposition. Fig. 6.14 shows SEM and XTEM images of a contaminated Ge nanowires/Si sample after laser irradiation using the LTO capping layer. The contamination is believed to be hydrocarbon and other materials from SEM and FIB scanning. A highly defected layer is formed at the surface. Further experiments are needed to more fully understand and control the LTO deposition on the Ge nanowires/Si structures.

The surface of the substrate at the SiGe nanowire location in Fig. 6.12 is modified. Two shoulders are formed at the sides of the SiGe nanowire. This surface evolution is thought to be caused by the same reason as we discussed in section 6.2, in which case no LTO capping layer is used. However, the shape of the shoulder is less rounded and the size is much smaller. This indicates the LTO capping layer also effectively stabilizes the surface during the laser induced melting and resolidification process.

#### 6.4 Conclusion

In conclusion, defect-free SiGe nanowires/Si with a cross-section of ~  $25 \times 95 \text{ nm}^2$ are successfully fabricated using the pulsed laser induced epitaxy process using a thin LTO capping layer deposited on a Ge nanowires/Si sample. The SiGe nanowires formed in the 26 ns melting and recrystallization duration have spherical interfaces with the Si substrate. In the PLIE process, Ge diffuses into the Si substrate, mixes with the Si and forms the SiGe alloy. The quality of the LTO layer deposited on the Ge nanowires is believed to be important for the uniform diffusion of Ge into the substrate to obtain reproducible uniform SiGe nanowires. The LTO capping layer is also seen to prevent the excessive side diffusion of Ge into the Si substrate. It also stabilizes the molten surface of the substrate during the PLIE process for formation of the SiGe nanowires, resulting in less surface modification. The PLIE process for SiGe nanowires/Si samples fabricated without using the LTO capping layer results in excessive side diffusion of the Ge in the Si substrate, and is inappropriate for the fabrication of SiGe nanowires. Future work should focus on obtaining high quality LTO depositions on Ge nanowires/Si structures and the mechanism of Ge diffusion in the PLIE process of LTO/SiGe nanowires/Si structures.

#### 6.5 Reference

- [6.1] Yih Chang, J. Kramer, S.Y. Chou, T.W. Sigmon, A.F. Marshall, "Fabrication of patterned Ge<sub>x</sub>Si<sub>1-x</sub>/Si layers by pulsed laser induced epitaxy", *Appl. Phys. Lett.* Vol. 58, no.19, 2150(1991).
- [6.2] G.E. Ellison and F.A. Modine, Appl. Phys. Lett. 41, 180(1982).
- [6.3] J. Kolodzey, S. Aljishi, R. Schwarz, D. Slobodin, and S. Wagner, J. Vac. Sci. Technol. A 4, 2499(1986).
- [6.4] K.H. Weiner, Ph.D. Dissertation, Stanford University, Stanford, CA, 1989.
- [6.5] Y. Chang, Ph.D. Dissertation, Stanford University, Stanford, CA, 1991.
- [6.6] Jack McCarthy, personal communication; to be presented at MRS spring meeting, San Francisco, April, 1995.

#### Chapter 7

## **CONCLUSIONS AND FUTURE WORK**

### 7.1 Summary and Conclusions

In the preceding chapters we demonstrated that pulsed *uv* laser induced epitaxy is a simple and effective process for fabrication of high quality SiGe nanowires on a silicon substrate. We discussed potential device applications of these SiGe nanowires. Two different approaches, i.e. focused ion beam micromachining and e-beam lithography, for obtaining patterned Ge nanowires are studied in this work. Based on the results described in the preceding chapters of this report we present the following conclusions:

1) An advanced process is developed for fabrication of SiGe nanowires using pulsed laser induced epitaxy using a XeCl excimer laser operating at a wavelength of 308 nm. In particular, Ge nanowires with a cross-section of  $\sim 6 \times 60 \text{ nm}^2$  are formed in a lift-off process on the Si substrate using e-beam lithography, followed by a  $\sim 10 \text{ nm}$  low-temperature oxide deposition. Defect-free SiGe nanowires with a cross-section of  $\sim 25 \times 95 \text{ nm}^2$  are then produced by direct laser irradiation of the sample at an effective energy fluence of  $\sim$ 0.69 J/cm<sup>2</sup>, which corresponds to a Si melt duration of 26 ns. The low temperature oxide is used as a capping layer to prevent side diffusion of the Ge in the Si substrate therefore keeping the width of the SiGe nanowire formed near to its patterned width. Excessive side diffusion of the Ge resulting in subsequent broadening of the SiGe wire width is observed when the low-temperature oxide capping layer is not used. Since the low-temperature oxide partially reflects the laser beam, its quality and uniformity are important for uniform Ge diffusion, and in turn, the uniformity of the SiGe nanowires. 2) In the above mentioned PLIE process for fabrication of the SiGe nanowires, Ge diffuses through the thin native oxide beneath the Ge wires forming the SiGe alloy in the Si substrate. No evidence of Si migration into the Ge wire location is observed. Interaction of the laser beam with the Ge nanowires/Si structure produces surface evolution in the PLIE process. This surface evolution is attenuated by the use of the LTO capping layer on the Ge nanowires.

3) Fabrication of SiGe nanowires in Si is made possible by the availability of technologies to initially fabricate Ge nanowires on the Si substrate. E-beam lithography proved to be an effective technique for fabrication of the Ge nanowires. Focused ion beam micromachining is also a promising technique to fabricate Ge nanowires. However, problems resulting from unexpected removal of materials by the FIB tail and contamination from the enhanced etching agent need to be addressed. As patterning technologies advance, we believe Ge nanowires with widths of < 10 nm will be available in the near future, resulting in the smaller dimension for the SiGe nanowires.

4) An advanced FIB assisted, cross-sectional TEM specimen preparation technique is developed for examination of the localized surface nano-structures. In particular, a multilayer protection scheme i.e. Pt/Au-Pd/C/nano-structures/Si is developed with FIB assisted marking and *in situ* Pt deposition. This technique allows effective high resolution crosssectional TEM characterization of SiGe nanowires with a high success rate.

5) Two advanced device structures, a quantum wire device and a lateral SiGe HBT based on SiGe nanowire technology are proposed. The lateral SiGe HBT is designed based on gas immersion laser doping, a technology similar to PLIE. The associated processing procedures used are compatible with current advanced CMOS processes. A MED-ICI simulation of the lateral SiGe HBT shows a maximum cut-off frequency of 36 GHz

with a current gain of 260. Performance characteristics of the device are comparable to vertical SiGe HBT structures.

#### 7.2. Suggestions for Future Work

In this work, we demonstrate the feasibility of using pulsed laser induced epitaxy to fabricate high quality SiGe nanowires. The effort has focused on the development of the processes for fabrication of the wires, not the mechanism understanding PLIE or applications and electrical characteristics of the SiGe nanowires. Both are beyond the scope of this work and represent individual research topics for future efforts. As a result of our efforts, the following topics are considered to be of great importance for continued development of PLIE technology for application in nano-fabrication.

## 1) Improving the quality of SiGe nanowires fabricated using PLIE

In this work, we observed that the quality of the low temperature oxide used for the capping layer on nanowires formed by the lift-off process is important in determination of the quality of the SiGe nanowires. The deposition process of this low-temperature oxide on the Ge nanowires/Si structures should be improved in order to obtain high quality, uniform thin film LTO on the nano-structures. Since the perfection and dimensions of the SiGe nanowires depend upon the quality and dimension of the Ge nanowires, further investigation into the fabrication of minimum dimension, high quality Ge nanowires is required to achieve the minimum dimension and highest quality for the SiGe nanowires. In addition, all work must performed in a clean room environment to avoid contamination seen in this effort.

### 2) Further investigation of the mechanism of PLIE

At the present time, the mechanisms behind the PLIE process is not fully understood. We discussed this briefly in Chapter 2. In Chapter 6, we described, for the first

time, the surface evolution and Ge diffusion behavior observed during the PLIE process, but have little knowledge about the physics behind the phenomena. Further investigation into the mechanisms understanding the PLIE process is a topic of interest for the future applications of this technique.

## 3) Device demonstration and electrical characterization

The primary undertaking of this effort was to show the feasibility of fabrication of SiGe nanowires using PLIE. We also proposed two novel device structures based on the SiGe nanowires in Chapter 3. Fabrication of these devices and their electrical characterization is necessary to prove that PLIE technology may be applied in nano-fabrication. In addition, SiGe quantum dot structures may be demonstrated using the PLIE technology as an understandable further demonstration of the capability of PLIE in this arena.

l

an an an and

the second a piller of the local second